## Is Multi-Path Transport Suitable for Latency Sensitive Traffic?

Kiran Yedugundla<sup>a,\*</sup>, Simone Ferlin<sup>b</sup>, Thomas Dreibholz<sup>b</sup>, Özgü Alay<sup>b</sup>, Nicolas Kuhn<sup>c,d</sup>, Per Hurtig<sup>a</sup>, Anna Brunstrom<sup>a</sup>

> <sup>a</sup>Dept. of Computer Science, Karlstad University, Karlstad, Sweden <sup>b</sup>Simula Research Laboratory, Oslo, Norway <sup>c</sup>IMT Télécom Bretagne, IRISA, Cesson-Sévigné, France <sup>d</sup>Centre National d'Etudes Spatiales (CNES), Toulouse, France

### Abstract

This paper assesses whether multi-path communication can help latency-sensitive applications to satisfy the requirements of their users. We consider Concurrent Multi-path Transfer for SCTP (CMT-SCTP) and Multi-path TCP (MPTCP) and evaluate their proficiency in transporting video, gaming, and web traffic over combinations of WLAN and 3G interfaces. To ensure the validity of our evaluation, several experimental approaches were used including simulation, emulation and live experiments. When paths are symmetric in terms of capacity, delay and loss rate, we find that the experienced latency is significantly reduced, compared to using a single path. Using multiple asymmetric paths does not affect latency – applications do not experience any increase or decrease, but might benefit from other advantages of multipath communication. In the light of our conclusions, multi-path transport is suitable for latency-sensitive traffic and mature enough to be widely deployed.

Keywords: Internet, latency, multi-path communication, transport protocols, MPTCP, CMT-SCTP

17

18

19

20

21

22

23

24

25

27

29

30

31

32

### 1 1. Introduction

Live and interactive applications are sensitive to 2 latency, as the user experience is negatively affected 3 when data is delayed. For instance, freezing a live video just 1% of the video duration is sufficient to 5 turn away 5% of the viewers [1]. Similarly, a la-6 tency of 60 ms suffices to degrade user experience 7 in Internet gaming [2]. Multiple ways of improving 8 the user experience of latency sensitive applications 9 are active subjects of research. However, as far as 10 we know, a weakly explored area is to determine 11 whether utilizing all available network interfaces at 12 the end host could improve such experience. In 13 recent times, deployed devices such as tablets and 14 smartphones are often equipped with both Wireless 15 LAN (WLAN) and cellular 3G or 4G interfaces. 16

\*Corresponding author

Preprint submitted to Elsevier

Multi-path transmission has been proposed to guarantee better resilience to link failures and a better use of resources. For instance, consider a connection using two interfaces simultaneously; if one of the interfaces (or underlying links) fails, the transmission can simply continue over the other interface. In a single-interface scenario, the transmission would be stalled and maybe require a connection re-establishment. It has also been shown that simultaneous transmission of data over multiple interfaces can increase the throughput, due to capacity aggregation [?]. Even if multi-path protocols have been shown to be more resilient to link failures and able to aggregate capacity to provide increased throughput, the impact of using multiple paths on latency has not been thoroughly investigated.

This paper fills this gap by assessing whether multi-path approaches are suitable transport protocols for applications transmitting latency-sensitive traffic, e.g., video, gaming and web traffic. Recent efforts within the Internet Engineering Task Force (IETF) include designing Multi-path TCP (MPTCP) [3] extensions to TCP [4] to en-

Email addresses: kirayedu@kau.se (Kiran Yedugundla), ferlin@simula.no (Simone Ferlin), dreibh@simula.no (Thomas Dreibholz), ozgu@simula.no (Özgü Alay), nicolas.kuhn@cnes.fr (Nicolas Kuhn), perhurt@kau.se (Per Hurtig), annabrun@kau.se (Anna Brunstrom)

able end-to-end connections to span multiple paths 40

simultaneously. Similarly, Concurrent Multipath 41 Transfer for SCTP (CMT-SCTP) [5, 6, 7] is an ex-42 tension to the Stream Control Transmission Proto-43

col (SCTP) [8], enabling simultaneous multi-path 44 communication. We therefore evaluate their suit-45

ability to carry out latency sensitive traffic.

In our experiments we consider both symmetric 47 multi-path communication (e.g. WLAN-WLAN) as 48 well as asymmetric (e.g. WLAN-3G). For the actual 49 evaluations we use a combination of simulations, 50 emulations and real experiments to ensure a correct 100 51 assessment. 52

The remainder of this paper is structured as fol-102 53 lows. Section 2 presents an overview of CMT-SCTP 103 54 and MPTCP, and how these protocols solve the core 104 55 issues inherent in transport-level multi-path com-56 105 munication. Section 3 describes the applications 106 57 used in our evaluation and their latency require-107 58 ments. In Section 4, the experimental setup is de-108 59 tailed. Section 5 presents and explains the results 60 109 obtained. In addition to that, Section 6 provides 61 110 an in-depth discussion of the results. Section 7 dis-111 62 cusses related work on multi-path transport. Fi-112 63 nally, Section 8 concludes the paper and discusses 64 possible future work in this area. 113 65

#### 2. Multi-Path Transport 66

section introduces CMT-SCTP This and 117 67 MPTCP, the current key multi-path transport 118 68 protocols. The core issues of multi-path com- 119 69 munication, and how these are addressed by 120 70 CMT-SCTP and MPTCP, are then described. 71

#### 2.1. CMT-SCTP 72

SCTP [8? ] is a transport protocol originally 73 124 developed by the IETF Signaling Transport (SIG-74 125 TRAN) Working Group [9], as part of an architec-126 75 ture to provide reliable and timely message deliv-127 76 ery for Signaling System No. 7 (SS7) [10] telephony 128 77 signaling information, on top of the Internet Proto-129 78 col (IP) [11]. While motivated by the need to carry  $_{130}$ 79 signaling traffic, SCTP was designed as a general 131 80 purpose transport protocol on par with TCP [4] 132 81 and UDP [12]. While SCTP can offer functionality 133 82 similar to TCP, such as ordered and reliable trans-134 83 mission or congestion controlled transport, its op- 135 84 85 tions can be easily set so that SCTP rather features 136 unordered transmission or multi-homing. This flex-137 86 ibility is one main advantage of SCTP as opposed 138 87 to TCP. 88

The multi-homing feature of SCTP allows a single association (or connection) between two endpoints to combine multiple source and destination IP addresses. These IP addresses are exchanged and verified during the association setup, and each destination address is considered as a different path towards the corresponding endpoint. Using the Dynamic Address Reconfiguration protocol extension [13], it is also possible to dynamically add or delete IP addresses, and to request a primary-path change, during an active SCTP association.

While SCTP multi-homing [? 8] targets robustness and uses only one active path at a time, several researchers have suggested the concurrent use of all paths for sending data. Budzisz et al. [14] provides a survey of these approaches. In this paper, we consider the most complete of these proposals, Concurrent Multipath Transfer for SCTP (CMT-SCTP) [5, 6, 7]. CMT-SCTP improves the internal buffer management procedures of SCTP, transmission over multiple paths and reordering with its single sequence-number space. Assuming disjoint paths, CMT-SCTP applies the original SCTP congestion control [8] for each path independently.

### 2.2. MPTCP

89

90

91

92

93

94

95

96

97

98

99

101

114

115

116

121

122

123

139

Multi-Path TCP (MPTCP) [16] is a set of extensions to TCP [4, 15] developed by the IETF MPTCP working group [17] to enable simultaneous use of multiple paths between endpoints. The motivation behind MPTCP is more efficient resource usage and improved user experience through improved resilience to network failure and higher throughput.

To use the MPTCP extensions the initiator of a connection appends a "Multipath Capable" (MP\_CAPABLE) option in the SYN segment, indicating its support for MPTCP. When the connection is established, it is possible to add one TCP flow, or subflow, per available interface to this connection by using a "MPTCP Join" (MP\_JOIN) option in the SYN segment. Once the MPTCP connection has been fully established, both end hosts can send data over any of the available subflows.

While MPTCP transparently divides user data among the subflows, simultaneous transmission may cause connection-level packet reordering at the receiver. To handle such reordering, two levels of sequence numbers are used. Apart from the regular TCP sequence numbers that are used to ensure in-order delivery at subflow level, MPTCP uses a 64-bit data sequence number that spans the entire



Figure 1: Definition of a path as a sequence of links between a sender and a receiver

187

188

190

192

193

194

195

196

107

100

200

201

202

205

206

207

210

211

212

MPTCP connection and can be used to order data 140 arriving at the receiver. To ensure fairness [?] on 141 bottleneck links shared by subflows of a MPTCP 142 flow and other TCP flows, MPTCP extends the 143 standard TCP congestion control. Running ex-14 isting TCP congestion control algorithms indepen-145 dently would give MPTCP connections more than 146 their fair share of the capacity if a bottleneck is 147 shared by two or more of its subflows. To solve this 148 MPTCP uses a coupled congestion control [18] that 149 links the increase functions of each subflows' con-150 gestion control and dynamically controls the overall 151 aggressiveness of the MPTCP connection. The cou-152 pled congestion control also makes resource usage 153 203 more efficient as it steers traffic away from more 154 204 congested paths to less congested ones. 155

#### 2.3. Core Issues 156

This section presents the core issues that are re-157 208 lated to the use of multiple paths and how they are 209 158 addressed by CMT-SCTP and MPTCP. 159

#### 2.3.1. Path Management 160

As shown in Figure 1, a path is a sequence of links 161 213 between a sender and a receiver [3], over which it 214 162 is possible to open a subflow. A multi-path proto- 215 163 col must define a path management strategy. The 216 164 strategy needs to find suitable paths to open sub- 217 165 flows over and decide whether one or more subflows 218 166 should be opened over a specific path. For short or 219 167 extremely time-sensitive flows, the choice of path 220 168 for the initial connection establishment might be 221 169 very important. For example, if (i) two paths  $(p_1 222)$ 170 and  $p_2$ ) are available, (ii) both paths have the same 223 171 capacity and (iii) the RTT of  $p_1$ ,  $r_1$ , is significantly <sup>224</sup> 172 higher than the RTT of  $p_2$ ,  $r_2$ , (e.g.  $r_1 > 10 \times r_2$ ), 225 173 then whether the first subflow will be opened over 226 174  $p_1$  or  $p_2$  would seriously impact the latency. The 227 175

number of subflows to open over a path is a problem that is not very well studied. While the Linux implementation of MPTCP supports this using its ndiffports path manager, as described later in this section, it is often regarded as unnecessary to open more than one subflow per path as they typically would traverse the same links and compete for the same network resources. However, in some specific environments, e.g. datacenters, the network might conduct load balancing between subflows, routing them over disjoint subpaths. In such situations there might be benefits of creating several subflows per path, as shown in [19].

For CMT-SCTP a path is defined by the destination IP address and port number. To manage paths, CMT-SCTP employs a simple strategy where the association is established during a 4-way handshake in which available IP addresses are exchanged and verified. The handshake is conducted over the default interface of the host, and after its completion each destination address is considered as a path and implicitly also as an opened subflow. The interfaces of the hosts are pairwise connected over two different subnets, resulting in two possible paths. In addition to the establishment of subflows during the association setup, there is an extension to CMT-SCTP called Dynamic Address Reconfiguration (DAR) [13] which enables an end-host to dynamically add and remove IP addresses to an existing connection.

Like CMT-SCTP, MPTCP consults the routing table to determine which interface to initiate the connection over. During the establishment phase, realised by a 3-way handshake, IP address information are exchanged between the hosts in a fashion similar to that of CMT-SCTP. However, after connection establishment, MPTCP cannot make full use of the other host's IP address information and start sending data over all paths straight away. Notably, at this point it can only use the path used for the connection establishment. Additional subflows must be opened as separate TCP connections and joined to the MPTCP connection using the "MPTCP Join" option in their SYN segments. Another difference, as compared to CMT-SCTP, is the availability of multiple path managers in MPTCP. For example, the Linux implementation of MPTCP provides four different path managers: default, fullmesh, ndiffports and binder. Using the default path manager, a host does not advertise additional IP addresses but uses the other hosts advertised IP addresses to create new subflows. The full-mesh

strategy uses an opposite approach: all available IP 279 228 addresses are exchanged and used to open a sub- 280 229 flow over each and one of all the possible source- 281 230 destination IP address combinations. The ndiff-282 231 ports manager allows a user to open X subflows 283 232 over the default interface. Finally, the binder man-233 284 ager, implements Loose Source Routing as defined 234 285 in [20]. Similar to CMT-SCTP, MPTCP also in-235 286 cludes an extension to allow dynamic addition and  $_{\scriptscriptstyle 287}$ 236 removal of IP addresses. 237 288

#### 2.3.2. Scheduling 238

If multiple subflows are available, there are differ-239 291 ent ways to schedule the transmission of data. As 240 202 an example, a round-robin scheduler may iterate 203 241 over the available subflows and try to transmit an 242 294 entire congestion window over each subflow, while 295 243 another scheduler might only consider the "fastest" 244 available subflow. Scheduling in multi-path com-245 297 munication has therefore a large impact on the per-246 formance of data transmission. 247

One root cause for scheduling problems is the use 248 of paths with asymmetric characteristics. Figure 2 249 illustrates how asymmetric paths, in terms of RTT 250  $(RTT_2 = 10 \times RTT_1)$ , can affect data transmission. 251 Figure 2c, shows the so-called head-of-line blocking 252 problem. In this scenario, packets #3 and #4 (re-253 siding in the receiver's buffer) cannot be delivered <sup>304</sup> 254 to the application as packets #1 and #2 are still  $^{305}$ 255 in flight. Further, packets sent over the fast sub-256 flow can fill the receiver's buffer while waiting for 257 data transferred over the slow subflow. This issue is 258 known as receiver buffer blocking and is illustrated 259 in Figure 2d. 260

The usual scheduler in CMT-SCTP is a round-311 261 robin scheme targeting throughput maximization: 312 262 for every subflow in sequence, starting from the pri-313 263 mary one, it sends as much data over the subflow as <sup>314</sup> 264 the congestion window allows. As mentioned ear-315 265 lier, asymmetric paths can be problematic and this 316 266 is also true for CMT-SCTP. The problem is due to 317 267 a combination of the scheduling and occupancy of 318 268 the shared send and/or receive buffer space, and <sup>319</sup> 269 can cause the aforementioned problems of head-of- 320 270 line blocking and receiver buffer blocking. Detailed 321 271 classifications of the blocking issues are provided 322 272 in [21]. To remedy this problem, other schedulers <sup>323</sup> 273 have been proposed and developed for CMT-SCTP. 324 274 For example, chunk rescheduling [22, 7] is a mech-275 anism that re-injects the segment causing head-of-326 276 line blocking on a different subflow that has space 327 277 available in its congestion window. Furthermore, 328 278

Delay-Aware Packet Scheduling (DAPS) [23] is a scheduler that, given the RTT of the different subflows, tries to send packet sequences over them in a manner that guarantees in-order delivery at the receiver.

Similar to CMT-SCTP, several schedulers have been proposed for MPTCP. In the Linux implementation the default scheduler always tries to transmit data over the subflow with the shortest RTT, as long as there is free space in the congestion The default scheduling also includes a window. mechanism called Retransmission and Penalization (RP). This mechanism is similar to CMT-SCTP's chunk rescheduling and re-injects segments causing head-of-line blocking in a different subflow. In addition to the default scheduling mechanism, a weighted round-robin scheme is also available. The schedulers available for Linux have all been evaluated and compared in [24], identifying the shortest-RTT scheduler as the most successful in terms of throughput performance. The different schedulers are detailed in [25].

289

290

296

298

299

300

301

302

303

306

307

308

309

310

### 2.3.3. Congestion Control for Multi-Path Transport

When using multiple paths for transmission, different subflows cannot share a single congestion window, as each subflow is likely to have different characteristics and levels of congestion. There are, however, situations in which the subflows actually do share a bottleneck and thus have the same level of congestion. In such scenarios there must be some kind of collaboration between the congestion controllers of each subflow to ensure that the transport does not achieve more than its fair share of the network resources.

For CMT-SCTP there is no default congestion control mechanism that manages the transmission based on the combined congestion state of the subflows. This is likely due to an initial design assumption that subflows do not share bottlenecks. As discussed above, such an assumption is not always true, making CMT-SCTP potentially unfair to other traffic in the network. This problem has been addressed by several researchers and coupled congestion controllers have been proposed. Examples include e.g., CMT/RPv1 [26] and CMT/RPv2 [27].

The problem of not considering shared bottlenecks was addressed already in the design phase of MPTCP. The reason to why coupled congestion control should be used, the benefits of using



(a) At  $t_0$ ; asymmetric path  $RTT_2 = 10 \times RTT_1$ .



(c) At  $t_2 = t_1 + RTT_1$ ; #3 and #4 are received but can not be read because #1 and #2 are missing  $\Rightarrow$  head-of-line blocking.



(b) At  $t_1 = t_0 + \epsilon$ ; #1 and #2 on path 1; #3 and #4 on path 2.



(d) At  $t_3 = t_1 + 2 \times RTT_1$ ; #1 and #2 are not received and the receiver's buffer can not receive #7  $\Rightarrow$  receiver buffer blocking.

Figure 2: Head-of-Line Blocking and Receive Buffer Blocking

358

359

it, and what goals it has to achieve are all docu- 349 329 mented in [18]. To achieve these goals, various cou-330 pled congestion control schemes have been proposed 351 331 for MPTCP. These include the Linked-Increases 352 332 Algorithm (LIA) [18], the Opportunistic Linked- 353 333 Increases Algorithm (OLIA) [28] and the BALanced 354 334 Linked Adaptation (BALIA) [29]. At the time of 335 writing, the default congestion control in the Linux 336 356 MPTCP implementation is LIA. 337

### 338 2.3.4. Handling Loss and Retransmissions

When data is lost in multi-path transmission the 360 339 protocol must decide whether to retransmit this 361 340 data over the same subflow or over a different one. 362 341 CMT-SCTP features several schemes for retrans-363 342 mitting data, all detailed in [30]. A CMT-SCTP 364 343 sender maintains accurate information about the 365 344 345 working paths, as new data are transmitted over 366 every available subflow concurrently. Therefore, 367 346 many distinct strategies can be used. For exam- 368 347 ple, retransmitting lost data over the same subflow, 369 348

over the subflow with the largest slow-start threshold or using the subflow with the largest congestion window. While there is no default retransmission strategy for CMT-SCTP, we consider retransmissions over the subflow with lowest RTT to give latency-sensitive applications a benefit.

In MPTCP, the loss detection is performed at two levels: subflow level and MPTCP level. While loss typically is detected on subflow level, different strategies can be taken depending on how the loss was detected. If the loss is detected by the fast retransmit algorithm, data is only retransmitted over the same subflow. If, on the other hand, the loss is detected by an expiration of the RTO timer, the data can be retransmitted over both the same subflow and over an additional subflow chosen by the scheduler. The rationale for using different retransmission approaches depending on how the loss was detected is straightforward; fast retransmit only detects loss if feedback from packets sent after the lost packet(s) arrive at the receiver. Therefore, it is safe

to assume that no massive congestion event or link 418 370 breakage has happened, and that a retransmission 419 371 will arrive safely at the receiver. If no feedback is 420 372 received, however, the RTO will eventually expire, 421 373 and it is then safer to retransmit the lost packet(s)422 374 over both paths in case the path over which the 423 375 original transmission occurred is experiencing ma-376 424 jor congestion or other serious problems. 425 377

#### 3. Applications and their Requirements for 378 **Multi-Path Transport** 379

Traditionally, Internet has been dominated by 380 web traffic running on top of short-lived TCP con-381 432 nections [31]. For example, Ciullo et al. [32] found 382 that approximately 95% of the client TCP flows 383 and 70% of the server TCP flows were less than 384 10 segments. Although web traffic still constitutes 385 a large fraction of all traffic, video traffic and gam-386 ing traffic are now becoming more common. Recent 387 measurements [33] show e.g. that more than 53% of 388 the downstream traffic in North America is video 389 streaming. Forecasts (cf. [34]) also show that In-390 ternet video and gaming will continue to grow with 391 an annual compound growth rate of 29% for video 392 traffic and 22% for gaming. 393

Although the aforementioned traffic classes dif-394 fer significantly in many ways, they have a com-395 mon property – sensitivity to latency. In this paper 396 we will therefore use video, gaming and web traffic 397 to assess whether multi-path protocols are suitable 398 for latency-sensitive applications. The remainder 448 399 of this section describes the main characteristics of 449 400 the applications and discusses their requirements. 401

#### 3.1. Video Streaming 402

There are two main use-cases of video streaming: 403 453 Video on Demand (VoD) which is not broadcast live 404 and therefore do not have stringent latency require-454 405 ments; and direct live video which is broadcast live 455 406 and have requirements of low latency. 407

456 VoD applications has complete knowledge of the 408 457 content to transfer, and can therefore adapt the 409 sending rate appropriately. The quality of experi-410 ence of VoD is therefore less vulnerable to one-way 411 459 delay variations than the quality of experience of 412 460 direct live video. As the rationale of this work is to 413 461 414 assess whether multi-path transport protocols can be used for latency sensitive applications, we will 415 focus on direct live video as it is more sensitive to 416 latency. 417

The direct live video can be divided into two sub-categories: live broadcast of TV such as BBC iPlayer<sup>1</sup> and private video communications such as Skype<sup>2</sup>. We have focused on the latter category as such applications typically are interactive in their nature and thus more sensitive to latency. A1though Skype is proprietary and its communication protocol is closed, and may change over time, we use Skype-like video traffic in our evaluation. We do this for several reasons. First, Skype is a widely used application. Actually, Skype generates almost two percent of the total aggregate traffic in European fixed networks [?]. Second, although Skype mainly tries to use UDP for communication NATs and firewalls often force it to use TCP, making it an interesting use case for our experiments with multipath and reliable transports. Finally, Skype traffic is well studied and traffic characteristics have been reported by several researchers, making it relatively easy to model. According to [35, 36], it dynamically adapts its sending rate to the network conditions, with a frame rate per second going from 5 frames/s to 30 frames/s and a video bit rate from 30 Kbit/s to 950 Kbit/s.

**Requirements:** The latency requirements for a good user experience when considering live video communication are: one-way delay should be lower than 150 ms [37] and the difference in delay between packets (jitter) should be lower than 30 ms [37].

### 3.2. Gaming Traffic

426

427

428

429

430

431

433

434

436

437

440

441

443

444

445

447

450

451

452

Online gaming is often categorized into three different classes [38] each of them being characterized by specific traffic, as detailed in [39]. The classes are:

- first person avatar, e.g. First Person Shooter games (FPS)
- third person avatar, e.g. Massive Multiplayer Online games (MMO)
- omnipresent, e.g. Real Time Strategy games (RTS)

FPS games are tolerant to loss but are very delay sensitive, therefore they often use UDP as transport. MMO games, on the other hand, are less loss tolerant and require less bandwidth compared

<sup>&</sup>lt;sup>1</sup>http://www.bbc.co.uk/iplayer/live/bbcnews/ <sup>2</sup>https://www.skype.com/



Figure 3: Distribution of Packet Inter-Arrival Time and Packet Size for WORLD OF WARCRAFT [39]

to FPS games, therefore, a mixture of TCP and 462 490 UDP is used for transmission. TCP traffic of MMO 463 500 games is composed of multiple thin TCP flows. 464 501 Thin flows are characterized by a low transmission 465 502 rate where the majority of packets are much smaller 466 503 than the maximum transmission unit (MTU). An 467 50/ example of the traffic characteristics between the 468 505 server and a client of an MMO game is illustrated 469 506 in Figure 3. For RTS games, interestingly, latency 470 507 has a negligible effect on the outcome of the game, 471 508 indicating that RTS game-play clearly favors strat-472 egy over the real-time aspects [40]. 473

Considering the popularity of MMO games [39], 474 and the fact that they use TCP, this paper assesses 475 whether there are any benefits in using multiple 476 paths at the transport layer to carry the traffic gen-477 erated by an MMO game entitled Age of Conan. 478

**Requirements:** The requirements for a good 479 gaming experience highly depend on the class of 480 the game and the particular game itself. However, 481 low latency (lower than 60 ms is indicated in [2]) 482 and a small delay variation [41] are important for a 515 483 good gaming experience. 484

#### 3.3. Web Traffic 485

Figure 4 illustrates the distribution of common 519 486 web site sizes. In our experiments, to be represen-520 487 tative of the web, we have selected three web sites of  $_{521}$ 488



Figure 4: Distribution of Web Page Sizes according to [43]

different sizes: small (72 KiB), medium (1024 KiB) and large (3994 KiB). We also use software that emulates the behavior of a real browser, downloading the web sites using 6 concurrent connections over HTTP/1.1. More details on the web traffic can found in Section 4.3.3.

**Requirements:** The quality of user experience when accessing a web page is highly linked to the download completion time. For example, in [42], the authors report that Google measured that "an additional 500 ms to compute (a web search) [...] resulted in a 25% drop in the number of searches done by users.". Although the download completion time may not be the most relevant metric for modern browsers, as they often start rendering pages before completion, it is the most suitable metric to use when evaluating transports as it is "browser agnostic" and therefore neutral. For good web browsing experience, the download completion time is required to be as low as possible.

### 4. Experiment Setup

496

497

509

510

511

512

514

516

517

518

This section describes the experiment setup used for the performance evaluation of our target applications. The protocol implementations and network models used in the evaluations are also introduced.

### 4.1. Evaluation Tool Sets

In this study, we focus on the default and latest versions of the protocols. Our choice of evaluation tool sets, presented in this section, has been made on the basis of availability of source code and the fact that we wanted to consider both controlled and real life experiments.

4.1.1. Simulations, CMT-SCTP using OMNeT++ 568 522 There is no stable implementation of CMT-SCTP 569 523 for FreeBSD or Linux. Therefore, we could not per- 570 524 form emulations or real experiments using CMT- 571 525 SCTP. Instead, we performed simulations using 572 52 OMNeT++ [44] version 5.0b1 with the CMT-SCTP 573 527 model [7, 45, 46], and the NetPerfMeter applica- 574 528 tion model [47, 7] in the latest version of the INET 575 529 Framework [48], using the simulation processing 576 530 tool-chain SimProcTC [? ? ]. For the web traf- 577 531 fic simulations in Section 5.3, the HttpTools [?] 578 532 models provided as part of the INET Framework 579 533 are used. It was only necessary to add SCTP 580 534 support. The complete INET Framework sources 581 535 branch used for this paper is available online<sup>3</sup>. Most 582 536 changes have already been merged upstream. 583 537

Although there is an implementation of CMT- 584 538 SCTP available for NS-2 [5], it is unmaintained 585 539 and as of spring 2016 it is fairly out of date. The 586 540 OMNeT++ implementation, which is used for our 587 541 evaluations, includes the latest improvements and 588 542 options for SCTP and is therefore representing the 589 543 state-of-the-art in SCTP features. 590 544

### 545 4.1.2. Emulations, MPTCP in a Controlled Envi-546 ronment, using CORE 593

591

607

Because we wanted to evaluate MPTCP in a con-594 547 trolled environment, we ran experiments using the 595 548 Linux MPTCP implementation and the Common 596 549 Open Research Emulator (CORE) [49]. CORE en-550 ables the use of real protocols and applications to-551 597 gether with emulated network links, making the 552 598 evaluation of MPTCP easy to control and repli-553 cate. The Linux kernel implementation is the most 599 554 complete MPTCP implementation available, so this 600 555 setup also allowed us to use the most feature com-601 556 602 plete version of MPTCP. Using the same MPTCP 557 implementation for both the controlled experiments 603 558 in CORE and the real life testbed experiments also 604 559 allowed us to more easily compare and validate the 605 560 606 results. 561

# 4.1.3. Experiments, MPTCP in a Real-Life Envi ronment, using NorNet

Because we may drive biased conclusions if the protocols were to be evaluated only in controlled environments, we also assessed their performance using an environment where the network is used by 613 many other applications than the one we introduce in the network.

In order to realize this, we performed real network experiments on a dedicated testbed, namely NorNet Edge (NNE) [50]. NNE is a multi-homed testbed where each node is connected to multiple UMTS operators via Huawei E392-u12 modems as well as WLAN network. More specifically, in our evaluations, we consider one operational UMTS Mobile Broadband (MBB) network in Oslo, Norway. It is labelled as "3G". The WLAN access point is a public WLAN hotspot, connecting around 100 people during work hours in a large office complex with several interfering WLAN networks. The two WLAN networks used for the WLAN-WLAN scenarios are using the same technology (IEEE 802.11ag) and sharing the same medium, therefore certain level of interference is highly likely depending on the number of users and traffic patterns. We believe, this setup reflects a realistic scenario where the users cannot control these factors. The downside of this testbed experiments is that there might be a statistically insignificant and uncontrolled behavior for few packets. Furthermore, in this paper, we focus on the transport layer, therefore we study how transport layer reacts to such realistic path characteristics. In this real-world environment, MPTCP was tested under different scenarios for different applications as in the emulation setup.

### 4.2. Configuration of MPTCP and CMT-SCTP

Both MPTCP and CMT-SCTP have open source implementations, making it possible to enable and/or disable specific features. Due to readability we have chosen to only present the most important features, and their settings, in this section. Additionally, Table 1 provides a short summary of this information. For a full description of the protocol configurations and for all the experimental scripts and data, please see [51].

For MPTCP, we used the state-of-the-art Linux MPTCP implementation  $(v0.89.3)^4$ . We use the default options of MPTCP, including e.g. receive buffer optimization and coupled congestion control, with an exception for the Nagle algorithm, which is turned off in all application scenarios. Turning off Nagle is common practice when running applications that require low latency [?].

<sup>&</sup>lt;sup>3</sup>https://github.com/dreibh/inet/tree/ td-netperfmeter-for-integration.

<sup>&</sup>lt;sup>4</sup>Linux MPTCP: http://www.multipath-tcp.org.

The simulation uses the CMT-SCTP model for 660 615 OMNeT++ that is fully described in [7, 45]. As op- 661 616 posed to MPTCP, no default options are given for 662 617 CMT-SCTP. The latest version of the SCTP sim- 663 618 ulation model [46, 45] for OMNeT++ is used, im-664 619 plementing SCTP according to RFC 4960 [8] with 620 665 all state-of-the-art features and extensions. In ad-621 666 dition to the settings listed in Table 1, CMT-SCTP 667 622 was configured with the following features: 668 623

- burst mitigation with MaxBurst=4 (default 669 624 from [8, Section 15]) with "Use It or Lose 670 625 It" [52] strategy (i.e. behavior like the FreeBSD 671 626 SCTP implementation [46]); 672 627
- buffer splitting [22, 21] to avoid buffer blocking 674 628 issues [7, Section 7.5]. 629

All data is sent in SCTP/MPTCP messages of 677 630 up to 1,452 bytes (resp. 1,428 bytes), which corre- 678 631 sponds approximatively to full packets (including 679 632 headers) of 1,500 bytes. In CMT-SCTP, the size 680 633 of the payload depends on the number of chunks 681 634 that are gathered in one message, therefore the full 682 635 packet sizes may vary depending on the application 683 636 profile. 684 637

As explained in Section 2.2.1, (i) with CMT- 685 638 SCTP, one subflow can be opened on each work-686 639 687 ing path as soon as the 4-way handshaking process 640 has been operated and the primary path (i.e., the 641 689 first path on which data is transmitted) has to be 642 defined; (ii) with MPTCP, the first subflow that 690 643 is opened depends on the parameterization of the 691 644 Linux default interface. In our evaluations, when 692 645 the paths are homogeneous (i.e., WLAN-WLAN 693 646 or 3G-3G), the path on which the first subflow is 694 647 opened is chosen randomly; when the paths are het-695 648 erogeneous (i.e., WLAN-3G), the WLAN path is 696 649 used for the first subflow. 650

#### 4.3. Application Traffic Generation and Metrics 651

This section presents how we generate video 652 streaming, gaming and web traffics. The rationale 653 for using these applications and the characteristic 654 of the traffic that they generate are detailed in Sec-655 tion 3. 656

#### 4.3.1. Video Traffic 657

In this paper, we have not considered Video on 708 658 Demand traffic since it is hard to accurately model 709 659

or emulate this traffic in the various cases (emulation, simulation, experimentation) used in this article. Moreover, these applications are not interactive and might be seen as file transfer applications. Therefore, we considered Direct Live Video applications due to their delay sensitive nature. In order to generate Skype-like traffic we considered a constant bit rate application generating 950 Kbit/s with 30 frames/s.

### 4.3.2. MMO Gaming Traffic

673

675

676

For gaming traffic, we considered a set of trace files from the Massively Multiplayer Online Game Age of Conan, provided by Funcom [53]. These traffic traces extend over a very long time period. As it is extremely difficult and tedious to replay all the traces completely, we selected a set of three traces with a duration of 10 minutes each. The selection of traces was based on the possible full game play being captured in the trace. Full game play constitutes initial loading of game settings, player interaction and infrequent chunk updates depending on the game. All the traces contain a huge chunk of game setup data in the beginning of the connection followed by occasional small bursts of MTU-sized packets and small packets for the rest of the time. The selected traces were replayed using the D-ITG [54] traffic generator. In this process D-ITG is loaded with full trace and it generates packets of exact size and time sequence as seen in the trace file. This is a way of providing trace input to the experiments than generation based on a statistical setting. From here on, the traces are named Trace 1, Trace 2 and Trace 3. They have average packet inter-departure times of 181.4 ms, 74.1 ms, and 167.7 ms respectively. Furthermore, the average packet sizes are 142.7 bytes, 113 bytes, and 101.7 bytes respectively.

### 4.3.3. Web Traffic

697

698

699

700

701

703

704

705

706

707

As previously mentioned, we consider three classes (small, medium, large) which are representative for real web sites. The classes and sites are presented in Table 2. For the experiments, we stored the files from the three sites (Wikipedia, Amazon and Huffington Post) on a local server. Each website data contained different umber of objects of different sizes. Of the three, Wikipedia content was the smallest followed by Amazon and Huffington Post. The data stored in the local server was requested and downloaded from a client with 6 concurrent connections.

	Path management Section 2.2.1	Scheduling Section 2.2.2	Congestion control Section 2.2.3	Handling loss Section 2.2.4
MPTCP	full-mesh	<ul><li>(i) LowRTT</li><li>(ii) Retransmission and penalization</li></ul>	coupled (OLIA)	<ul><li>(i) Fast retransmit</li><li>on the same subflow</li><li>(ii) RTO on a subflow</li><li>chosen by the scheduler</li></ul>
CMT-SCTP	one subflow per working path	<ul><li>(i) packet-based round-robin</li><li>(ii) chunk rescheduling</li></ul>	uncoupled (NewReno)	retransmission on lowest RTT path

Table 1: Options for MPTCP and CMT-SCTP

Size	Domain name	Number of objects	Size of objects
small	Wikipedia (www.wikipedia.org)	15	72 KiB
medium	Amazon (www.amazon.com)	54	1024 KiB
large	Huffington Post (www.huffingtonpost.com)	138	3994 KiB

Table 2: Web Traffic Generation

742

743

762

764

767

769

4.4. Background Traffic Generation 710

#### 744 The congestion level in a network has a signifi-711 745 cant impact on the behavior of protocols employ-712 746 ing congestion control [55]. We therefore conduct 713 747 experiments both with and without background 714 748 traffic. Background traffic is generated with Net-715 749 PerfMeter [56, 47] as a mix of TCP and UDP flows 716 750 constituting one long TCP flow and 4 UDP on-717 751 off flows. The TCP flow has a saturated sender 718 752 sending as much data as possible with frame size of 719 753 1,460 bytes. Each UDP flow generates Pareto on-720 754 off traffic with shape 1.5 and scale 0.166667, send-721 ing 25 frames per second each of size 5,000 bytes. 722 The aggregate usage of UDP background flows were 723 757 maintained at 10% of the bottleneck link capacity 724 758 to be realistic [57]. The UDP flows carry data at 725 an average of 500 Kbit/s each in the WLAN-WLAN 726 759 scenario and 100 Kbit/s each in the 3G-3G scenario. 727 In each run, the background flows start before the 760 728 761 foreground experimental traffic and end after the 729 experimental traffic. 730

However, in the NorNet experiments even run-731 ning one experiment with the background traffic, 732 765 especially for online gaming and video streaming, 733 766 can eat up all the monthly data quota. Therefore, 734 we decided not to generate background traffic due 735 768 to the limited data quotas for the 3G subscriptions 736 and in order to provide consistent results, we have 737 run all NorNet experiments without background 738 traffic. 739

#### 4.5. Network and System Characteristics 740

#### 4.5.1. Topology 741

Figure 5 shows the topology used in our evaluations. The same topology was used for the simulations, emulations, and real experiments in Nor-Net. To understand the basic performance of protocols and highlight their characteristics, a simple topology is more useful than a complex topology. Though the topology can be seen as more general, this type of basic topology is also common practise for evaluating transport protocols, see for instance Common TCP Evaluation Suite  $^{5}$ . As shown in the figure, there are two paths on the client side and a single path on the server side. In the simulations and emulations the paths that connect the server and client are, of course, modeled. For the experiments to be realistic we used a parameterisation of the paths that is based on measurements that were conducted over NorNet prior to the evaluation.

### 4.5.2. Path Characteristics

Table 3 shows the capacity, end-to-end delay and packet loss rates of the WLAN and 3G paths that have been measured in the experimental testbed described in Section 4.1. The WLAN links are IEEE 802.11ag. The loss rate is what is experienced on the transport layer (e.g. datagrams), therefore it is the loss ratio after all link layer re-transmission schemes of underlying networks. Path 1 and Path 2 in Figure 5 will be assigned with these characteristics depending on the technology. In Table 3, we

<sup>&</sup>lt;sup>5</sup>https://tools.ietf.org/html/draft-irtf-iccrg-tcpeval-01



Figure 5: Topology

794

795

796

797

798

799

801

802

803

804

805

806

807

809

810

811

812

813

820

821

822

823

824

825

826

	WLAN	3G
Capacity [Mbit/s]	20-30	3 - 5
Propagation Delay [ms]	20 - 25	65 - 75
Loss $[\%]$	1 - 2	0
Homogeneous (WLAN)		
Path 1	х	
Path 2	х	
Homogeneous (3G)		
Path 1		х
Path 2		х
Heterogeneous		
Path 1	х	
Path 2		х

Table 3: Path Characteristics and Scenarios

also detail the three combinations of paths over 770 which the multi-path protocols are evaluated, that 771 are homogeneous WLAN (two WLAN paths), ho-772 mogeneous 3G (two 3G paths) and heterogeneous 773 (one WLAN and one 3G path). 774

#### 4.5.3. Buffer Sizes 775

814 System characteristics of the source and desti-776 nation are known to impact the end to end perfor-815 777 mance of the flows. In order to emulate the realistic 816 778 network scenarios, we use the system settings close 817 779 818 780 to the standard settings for respective technologies. The TCP buffer sizes (send buffer/receive buffer) 819 781 are set to be equivalent to the default Android set-782 tings, that are configured as follows: 783

- Homogeneous (3G): 256 KiB/256 KiB. 784
- Homogeneous (WLAN): 1024 KiB/2048 KiB. 785

Heterogeneous (WLAN-3G): 786 1024 KiB/2048 KiB. 787

Based on estimations from early measurement in 788 the NorNet testbed, performed during the planning 828 789 phase of the work, the queue lengths at each inter- 829 790 face of the router (see Figure 5) are set to 100 pack-830 791 ets for WLAN and 3750 packets for 3G. 792 831

Note. that the 3Gbuffer setting of 256 KiB/256 KiB prevents an overly large bufferbloat [58, 59] in a 3G/3G setup, while the setting of 1024 KiB/2048 KiB will make such a bufferbloat in the WLAN-3G case possible. We will explain this in detail with the results in Section 5 and particularly in Section 5.1.1.

#### 5. Experiment Results 800

This section presents the experimental evaluation and its results. The protocols are first evaluated through simulations and emulations in controlled environments, to identify the impact of various network parameters. This evaluation is then complemented with measurement results from a real environment.

For each experiment scenario, SCTP is compared to CMT-SCTP, and TCP Cubic is compared to MPTCP. For the homogeneous cases, we consider the average delay using TCP and compare it with that of MPTCP. We assume that we only have information about the technologies used. That is, for a WLAN-WLAN case, the WLAN channels might have different characteristics in terms of loss and delay, but this information is not available to the The user will most likely pick one of the user. WLANs randomly. Therefore, we consider the average WLAN TCP delay performance and compare it with the MPTCP delay performance. However, for the WLAN-3G scenario, the user will most certainly choose WLAN, since it has low delay, high capacity and is probably cheaper to use. Therefore, we compare the MPTCP delay with the TCP delay of WLAN. The evaluation of SCTP and CMT-SCTP is conducted in the same fashion.

### 5.1. Video Streaming

First, we evaluate video traffic performance for homogeneous and heterogeneous scenarios, considering both competing and non-competing traffic as explained in Section 4.3. We use application layer message delay as the performance metric for the transport protocol latency performance.

884

885

886

### 834 5.1.1. CMT-SCTP Simulations

Figure 6 presents the average message delays and 835 887 the variation in these delays in the form of box 836 000 plots [? for video traffic as described in Sec-837 880 tion 4.1. First, we consider the average SCTP mes-838 890 sage delay over 128 runs for video traffic without 839 891 any competing traffic and illustrate the results in 840 892 Figure 6a. Table 4 presents the percentage of traffic 841 803 sent over the different paths in CMT-SCTP. Note, 842 894 that although SCTP uses a certain primary path 843 895 for payload data transport, there is always a small 844 896 amount of control traffic (here: mainly heartbeats 897 845 to check the path status, see [8]) on the other path 846 808 as well. 847

In the homogeneous scenarios (WLAN-WLAN 848 900 and 3G-3G) we observe different behaviors for 849 901 WLAN-WLAN and 3G-3G cases. In the WLAN-850 902 WLAN scenario, with two similar WLAN paths, 851 003 multi-path transport leads to increased latency, and 852 00/ also increased latency variation, mainly due to the 853 005 reordering caused by retransmissions on the lossy 854 906 paths (i.e., 1%-2% packet loss; see Table 1). In the 855 907 3G-3G scenario, where both paths are lossless, the 856 908 performance of SCTP and CMT-SCTP are virtu-857 909 ally the same. 858 910

For the WLAN-3G scenario, we observe that 911 multi-path transport leads to increased delays and 912 delay variation. The main reason for this is the 913 reordering caused by the heterogeneous path char-914 acteristics. 915

For CMT-SCTP, it is important to note that 916 864 SCTP has its origins as transport protocol for sig-917 865 naling systems (see Section 2.1.1), where networks 918 866 are designed for specific applications. Therefore, 919 867 the most important implementations – the OM-920 868 NeT++ simulation model as well as CMT-SCTP 921 869 implementation in FreeBSD – currently only pro- 922 870 vide a very simple scheduler; data is scheduled on 923 871 the paths in a round-robin fashion. The intention 924 872 of using this scheduler is to improve throughput, 925 873 without caring about path delays. That is, once 926 874 data to send is available, and a path's congestion 927 875 window allows to send it, as much as possible is sent 928 876 on this path. Then, for further data to be sent, the 929 877 next path is tried. This mechanism is tightly com-878 930 bined with the SCTP burst mitigation that limits 931 879 the amount of consecutive packets sent at once over 932 880 a path. Both SCTP implementations apply burst 933 881

mitigation by using the "use it or lose it" [52] strategy, with a setting of MaxBurst=4 (default from [8,Section 15]). That is, if a certain number of bytes  $\alpha$ is acknowledged by the receiver side the sender would be allowed to send up to  $\alpha$  new bytes. The limit of in-flight bytes is given by the congestion window. However, if a non-saturated sender does not fully utilize its allowance given by the congestion window, the congestion window is reduced to the number of in-flight bytes plus MaxBurst\*MSS. As a result, using MaxBurst=4, only up to 4 packets are sent on a path before the next path is used. The round-robin scheduler together with the burst mitigation results in an increase in the message delay for CMT especially when the paths are heterogeneous or when there exists loss.

Next, the performance of CMT-SCTP was evaluated in the presence of competing background traffic (see Section 4.4) and the corresponding average message delays are illustrated in Figure 6b.

In the WLAN-WLAN scenario, similar to the case without background traffic, multi-path transport leads to increased latency and latency variation due to the reordering caused by retransmissions. We also see that the background traffic has no visible impact in the WLAN/WLAN scenario due to the random packet loss on the path. As the background flows are experiencing loss, the TCP background flow backs off before it causes any noticeable congestion, and as the UDP background traffic is only 10% of the link capacity it also has very limited impact. Hence in the WLAN-WLAN scenario there was no noticeable effect of congestion losses or queuing delay on the foreground flows as compared to the corresponding scenario without background traffic.

For the 3G-3G scenario, also similar to noncompeting traffic, we observe no significant performance difference for CMT-SCTP as compared to SCTP. Note that in this scenario, the send and receive buffer sizes are 256 KiB (see Section 4.5.3), and the background traffic leads to an increased delay due to bufferbloat. We observe an average delay of around 800 ms when background traffic exists as compared to 80-90 ms when there is no competing traffic.

The WLAN-3G results differ quite much from the non-background results. Note that for WLAN-3G case, we have different buffer settings: a send buffer of 1024 KiB and a receive buffer of 2048 KiB (see Section 4.5.3). These buffer settings allow the queue on the 3G path to grow, causing a significant



(a) Without Background Traffic

(b) With Competing Background Traffic

Figure 6: Average Message Delay for CBR Video Traffic over CMT-SCTP

Traffic	Background	WLAN-WLAN		3G-3G		WLAN-3G	
	Dackground	WLAN	WLAN	3G	3G	WLAN	3G
Video on CMT-SCTP	X	17.5	82.5	6.3	93.7	16.4	83.6
	1	17.5	82.5	4.5	95.5	79.2	20.8

Table 4: Path 1 Traffic Share (in %) for CBR Video Traffic over CMT-SCTP

966

967

968

969

970

971

973

bufferbloat. Here, we observe the delay on the 3G 958 934 path jumping to values of almost 4 s, making any in-959 935 teractivity virtually impossible. Using CMT-SCTP 960 936 leads to a significant reduction of the delay – due to 961 937 the additional usage of the low-latency WLAN path 962 938 to values of around 1.8 s. However, this delay is 963 939 still much higher compared to the delay of SCTP 964 940 on the WLAN path. 941

#### 5.1.2. MPTCP Emulation 942

Figure 7a presents the average message delay for 943 video traffic in all the scenarios considered, i.e., 944 WLAN-WLAN, 3G-3G, WLAN-3G. Each plot rep-945 resents the data for 30 repetitions when no back-946 972 ground traffic was present. 947

In the WLAN-WLAN scenario, the path delay 948 difference between the two paths is small (i.e., 20ms 974 949 25ms; see Table 1). Thus, the main factor that 975 \_ 950 determines the average message delay is the link 976 951 losses. Loss on one path causes the scheduler to 977 952 push data on the other path and eventually ex- 978 953 ploiting the availability of multiple paths. If both 979 954 paths have similar delay and loss as per the setup, 980 955 data is sent on both paths causing traffic to oscil-981 956 late between paths. Such oscillation of traffic be- 982 957

tween paths is known as flapping. In some of the repetitions, we observed that flapping and losses caused data to arrive out-of-order at the receiver, resulting in increased delays. However, on average, we observed that MPTCP improves the delay performance and reduces the delay variation as compared to TCP. This is in contrast to the relationship between CMT-SCTP and SCTP discussed above. The difference is due to the fact that MPTCP uses a lowest-RTT scheduler (see Section 4.5.3) that moves the sending of data between the paths in a better way as loss occurs. The limitation imposed by MaxBurst in CMT-SCTP also results in a less suitable distribution of the data as compared to MPTCP where the congestion window stays larger.

In the 3G-3G scenario, there are no losses and all the data is sent over only one path. Hence, there is no performance differences between MPTCP and TCP. Table 5 provides some insights on the share of data over each 3G path. The delay difference between the paths is small, but still enough to make the scheduler use only one of the paths. Certain configurations start with a non-optimal interface as default, and in those cases the scheduler eventually



Figure 7: Average Message Delay for CBR Video Traffic over MPTCP in CORE Emulation

switches to the other path. This is evident in Ta- 1015
ble 5, where 0.16% of the data was sent on the other 1016
3G path. 1017

In the heterogeneous scenario (WLAN-3G), the <sup>1018</sup> 986 WLAN path clearly has a lower average delay than <sup>1019</sup> 987 the 3G path, even for small amounts of loss on the <sup>1020</sup> 988 WLAN link. The behavior of the default sched- 1021 989 uler ensures that MPTCP uses the path with low-<sup>1022</sup> 990 est RTT. However, the performance of MPTCP was <sup>1023</sup> 991 observed to be worse than that of TCP in this sce- 1024 992 nario. MPTCP uses both paths due to losses in the <sup>1025</sup> 993 WLAN, triggering transmission over the 3G path <sup>1026</sup> 994 which otherwise would not be used due to the large 1027 995 path delay differences. In the case of video traf- 1028 996 fic considered for the experiments, the data share 1029 997 shown in Table 5 should be identical to the packet 1030 998 share, due to the fixed size of the packets. The large 1031 999 amount of data transmitted on the 3G link provokes 1032 1000 head-of-line blocking and the resulting application-<sup>1033</sup> 1001 level latency prevents MPTCP from reducing the <sup>1034</sup> 1002 latency. 1003

To analyse the performance in the case of com-<sup>1036</sup> peting traffic, we considered experiments with back-<sup>1037</sup> ground flows as specified in Section 4.4. The results <sup>1038</sup> are presented in Figure 7b. Similar to the CMT-

SCTP case, the background traffic has negligible 1039 impact over the WLAN paths as the loss encountered by the background flows prevents congestion from forming. Hence there is no impact of background traffic on the WLAN-WLAN scenario.

<sup>1013</sup> In the 3G-3G scenario, there is an improvement <sup>1044</sup> <sup>1014</sup> in the performance of MPTCP which was not visi- <sup>1045</sup> ble without background traffic (or for CMT-SCTP). MPTCP has a less varying average message delay than TCP in this scenario mainly due to the use of multiple paths and the lowest-RTT scheduling: since the distribution of the traffic considers the delay of each path, it is affected by the current congestion level of each path. The MPTCP scheduler used both paths and the data distribution among the paths is 85/15% as shown in Table 5. The path with the shorter base RTT may not always be the best path as the background traffic builds up queues in the network, leading to a somewhat more even share of the data between the paths as compared to the scenario without background traffic.

In the WLAN-3G scenario, MPTCP increases the delay compared to single path TCP over WLAN as the data was split between asymmetric paths. The underlying issue is the same as when there is no background traffic, but the background traffic causes the delays to get larger. As in the CMT-SCTP scenario, in Figure 7b, the delays for MPTCP and for TCP over 3G is higher than those of the 3G-3G scenario, due to the larger receive buffers in the WLAN-3G scenario.

### 5.1.3. MPTCP Real Measurements

We have run over 30 experiments in the NorNet Edge (NNE) testbed for the video traffic and illustrated the delay measurements in Figure 8. Note that different from WLAN links, in mobile broadband networks, each user has a dedicated channel. Therefore, we assume that the user is only stream-

Traffic Bac	Background	WLAN-WLAN		3G-3G		WLAN-3G	
	Dackground	WLAN	WLAN	3G	3G	WLAN	3G
Video	X	54.19	45.81	99.84	0.16	51.96	48.04
VIGEO	1	55.67	44.33	85.24	14.76	51.98	48.02

Table 5: Video traffic data share per path using MPTCP

	WLAN-	WLAN	3G-	-3G	WLAN-3G		
	WLAN	WLAN	3G	3G	WLAN	3G	
Video	65.05	34.9	93.8	6.19	20.41	79.5	

Table 6: Video traffic data share per path using MPTCP with NorNet

1075

1076

1077

1078

1079

1080



Figure 8: Average Message Delay for CBR Video Traffic over 1081 MPTCP in NorNet Experiment 1082

ing video without running any other bandwidth de- 1083 1046 manding applications. 1047

For the homogeneous cases (e.g., WLAN-WLAN 1084 1048 and 3G-3G), we observe that the paths can have 1085 1049 quite different delay values although we are using 1086 1050 the same technology. This results in delay differ- 1087 1051 ences between the paths, compared to the emula- 1088 1052 tions, and impact the performance of MPTCP. For 1089 1053 example, in the 3G-3G scenario, we observe that de- 1090 105 lay with MPTCP lies between the delay with TCP 1091 1055 of the two 3G paths and MPTCP provide delay 1092 1056 values much closer to the 3G path with lower delay 1093 1057 with TCP. Similarly, for the WLAN-WLAN sce- 1094 1058 nario, the delay with MPTCP is on average closer 1095 1059 to the delay of the WLAN path with the lower TCP  $_{1096}$ 1060 delay. We further observe large variations in the de- 1097 1061 lay values among different experiment runs. This is 1098 1062 in fact an expected result in real networks where 1099 1063 the channel conditions can be very dynamic. These 1100 1064 observations can further be verified by looking into 1101 1065 1066 the video traffic data share tabulated in Table 6. 1102 For the heterogeneous scenario (WLAN-3G), we 1103 1067 observe that the delays achieved by MPTCP is 1104 1068

higher than the delay of TCP on the WLAN path, 1069 since MPTCP occasionally uses 3G path that has 1070 higher delay values as compared to WLAN. Al-1071 though we have higher variations in the experimen-1072 tal results, this behavior is in general consistent 1073 with the emulation results.

### 5.2. Gaming Traffic

Gaming traffic is the second application traffic type that we consider. We start by presenting the simulation results for CMT-SCTP and continue with emulation and live experimentation results for MPTCP. We use the gaming traffic presented in Section 4.3.2. Same as for video traffic, we use application layer message delay as the performance metric.

### 5.2.1. CMT-SCTP

The average SCTP message delays over 128 runs are presented in Figure 9 (without background traffic) and Figure 10 (with background traffic according to Section 4.4) for the three gaming traces (Trace 1, Trace 2 and Trace 3) described in Section 4.3.2. The traffic share between the paths (i.e. the first path) is provided in Table 7.

A particular property of the gaming traffic is its mix of traffic patterns due to the different phases of game play (see Section 4.3.2). The CMT-SCTP scheduler provides no gain for the small packets sent during the game. Therefore, when using symmetric WLAN paths in the WLAN-WLAN scenario, CMT-SCTP only leads to additional delay caused by reordering for the occasional small bursts of packets sent.

For the 3G-3G scenario, as explained previously for the video traffic, the small send and receive buffer settings of 256 KiB keeps bufferbloat and packet reordering small. Therefore, the effort for



Figure 9: Average Message Delay for Gaming Traffic over CMT-SCTP (without Background Traffic)



Figure 10: Average Message Delay for Gaming Traffic over CMT-SCTP (with Competing Background Traffic)

Traco	Background	WLAN-WLAN		3G-3G		WLAN-3G	
ITace	Dackground	WLAN	WLAN	3G	3G	WLAN	3G
Gaming T1	×	24.1	75.9	30.9	69.1	33.7	66.3
	✓	24.4	75.6	32.3	67.7	57.9	42.1
Caming T2	×	36.8	63.2	37.5	62.5	47.4	52.6
Gaming 12	✓	36.9	63.1	23.9	76.1	63.9	36.1
Gaming T3	X	36.4	63.6	37.3	62.7	48.4	51.6
	✓	36.5	63.5	22.9	77.1	65.0	35.0

Table 7: Path 1 Traffic Share (in %) for Gaming Traffic over CMT-SCTP

reordering messages remains small as well. How- 1156 1105 ever, due to the higher network latency, the burst 1157 1106 mitigation handling leads to some performance gain 1158 1107 with CMT-SCTP; the non-saturated sender does 1159 1108 not fully utilize its allowance given by the conges- 1160 1109 tion window. Therefore, the congestion window is 1161 1110 reduced by the burst mitigation. This limitation 1162 1111 keeps the congestion window small, allowing CMT- 1163 1112 SCTP to send messages more quickly on the two in- 1164 1113 dependent paths when small bursts of packets need 1165 1114 to be sent. 1115 1166

Again, as explained in Section 5.1 there is 1167 1116 high bufferbloat on the 3G path in the WLAN- 1168 1117 3G case, due to the send/receive buffer sizing of 1169 1118 1024 KiB/2048 KiB. This leads to a significant re- 1170 1119 ordering for CMT-SCTP during the initial large 1171 1120 burst of the game. Therefore, no significant per- 1172 1121 formance improvement is achieved when no back- 1173 1122 ground traffic is present. However, in the scenario 1174 1123 with background traffic, CMT-SCTP leads to some 1175 1124 improvement due to the distribution of traffic over 1176 1125 two paths. Nevertheless, the latencies caused by 1177 1126 the bufferbloat of at least 2 s make any gaming in- 1178 1127 teractivity impossible. 1128 1179

#### 5.2.2. MPTCP Emulation 1129

1130

1131

1132

1181 Figure 11 shows the delay values calculated over 1182 30 runs for each gaming trace and the distribution 1183 of the data on each path is shown in Table 8. 1184

1180

In the WLAN-WLAN scenario, it is clear that 1185 1133 one path is more used than the other, which was 1186 1134 not the case for the video traffic. This is due to the 1187 1135 limited amount of data to transmit for the many 1188 1136 small packets. Losses on the WLAN have minimal 1189 1137 impact on performance as there is less data to send 1190 1138 on the other path in the event of loss during large 1191 1139 portions of the game. In the 3G-3G scenario, the 1192 1140 average delay using MPTCP is similar to that of 1193 1141 TCP. We also observe a split of data over the paths, 1194 1142 which was not the case for the video traffic. 1143 1195

Though the scenarios WLAN-WLAN and 3G-3G 1196 1144 are symmetric in nature based on the characteris-1145

tics, one of the interfaces is the best in any given 1197 1146 configuration. The path delay settings are ran-1198 1147 domly drawn from a range of values (see Table 1). 1199 1148 The maximum possible difference between two path 1200 1149 delays are 5ms in WLAN-WLAN and 10ms in 3G-1201 1150 3G, respectively. This difference is sufficient for the 1202 1151 MPTCP scheduler to estimate, adapt and change 1203 1152 outgoing path for a packet. If the default interface 1204 1153 is the best of the two available interfaces, then the 1205 1154 flow uses mostly this interface. However, when the 1206 1155

default interface is not the best of the two available interfaces, the MPTCP scheduler will send the first few packets on the default interface before settling with the other interface. Due to the initial large burst and the long-tail nature of the gaming traffic, the first few packets that were sent on a possibly sub-optimal interface represents a large chunk of the data share, although most packets are transferred on the other interface. This also results in very similar performance for MPTCP and TCP.

It is also worth noting the difference to CMT-SCTP here, which showed a clear gain in the 3G-3G scenario. The difference for CMT-SCTP is that all paths are available for transmission immediately and it was able to spread the initial burst of data over both paths to reduce the delay. As MPTCP sets up the second subpath in parallel with starting the data transfer, and also uses a larger initial congestion window, there was no gain during the initial burst of data from the game setup.

In the asymmetric WLAN-3G scenario, the average MPTCP delay values are similar to the TCP delay of the WLAN path. This is due to most of the data being sent over the default (better) path, which in this case is the WLAN.

Figure 12 presents the average delay of MPTCP versus TCP for each gaming trace with competing background traffic. The impact of background traffic for gaming is very similar to the video traffic case. In the 3G-3G scenario, there is a lower and less variable average message delay using MPTCP, due to the scheduling. For the WLAN-3G scenario. MPTCP increases the delay compared to single path TCP over WLAN as some of the data is sent over the 3G path. Still, the degradation is smaller and the improvement in relation to TCP over 3G is larger than in the video traffic scenario. This as a smaller fraction of data is sent on the 3G path in this scenario as discussed above. Again, background traffic has no impact in the WLAN-WLAN scenario.

### 5.2.3. MPTCP Real Measurements

We illustrate gaming traffic delay performance for real-world measurements in Figure 15 and the traffic distribution over the paths is presented in Table 9. For the homogeneous scenarios, we observe that the average TCP delay is very similar to the MPTCP delay. There are small variations among different traces, where in one trace MPTCP's delay is a little bit lower than the average TCP delay and in one trace it is a little bit higher. We observed



Figure 11: Average Message Delay for Gaming Traffic over MPTCP in CORE Emulation (without Background Traffic)

Traffic	Background	WLAN-	WLAN-WLAN		3G-3G		WLAN-3G	
maine	Dackground	WLAN	WLAN	3G	3G	WLAN	3G	
Gaming T1	×	64.88	35.12	80.85	19.15	94.03	5.97	
	1	78.58	21.42	85.93	14.07	94.27	5.73	
Coming T2	×	76.79	23.21	83.40	16.60	97.98	2.02	
Gaming 12	1	67.65	32.35	93.42	6.58	99.23	0.77	
Gaming T3	X	78.30	21.70	92.01	7.99	99.29	0.71	
	1	69.33	30.67	87.76	12.24	99.31	0.69	

Table 8: Gaming traffic data share per path using MPTCP

Troffic	WLAN	30	G-3G	WLAN-3G		
ITAILIC	WLAN	WLAN	3G	3G	WLAN	3G
Gaming T1	77.67	22.32	0	100.0	40.74	59.25
Gaming T2	59.10	40.89	0	100.0	71.39	28.6
Gaming T3	54.57	45.42	0	100.0	30.31	69.69

Table 9: Gaming traffic data share per path using MPTCP in NorNet



Figure 12: Average Message Delay for Gaming Traffic over MPTCP in CORE Emulation (with Competing Background Traffic)

that the packet shares are slightly more split to- 1242
wards the better path in the emulations compared 1243
to the experiments. 1244

For WLAN-3G, we observe that MPTCP delay 1245 is slightly higher than the TCP WLAN delay. We 1246 observe that the packet share on the 3G path is 1247 higher in the experiments compared to the emula- 1248 tions, increasing the relative MPTCP delay slightly 1249 compared to the emulation results. Overall, the 1250 general trends seen between the protocols are still

<sup>1217</sup> similar in the experiments and in the emulations.

1251 1252 1253

1260

1261

Finally, we evaluate the latency of web traffic  $_{1254}$ for the homogeneous and heterogeneous scenarios,  $_{1255}$ both with and without background traffic. For the  $_{1256}$ web traffic, we chose the web site download time  $_{1257}$ as the metric for transport protocol latency perfor-  $_{1258}$ mance.  $_{1259}$ 

### 1225 5.3.1. CMT-SCTP Simulations

The web site download time results for the three 1262 1226 web site scenarios are presented in Figure 16 (with- 1263 1227 out background traffic; 256 runs) and Figure 17 1264 1228 (with background traffic according to Section 4.4; 1265 1229 1024 runs). Table 10 provides the corresponding 1266 1230 traffic share for the two paths. Clearly, the bene- 1267 1231 fit of CMT-SCTP usage increases as the web site 1268 1232 size grows. The Wikipedia site (see Table 2), hav- 1269 1233 ing only 72 KiB of payload data, is the smallest 1270 1234 of the three sites. Therefore, the benefit of using 1271 1235 CMT-SCTP for this web site is only small. 1272 1236 In the two scenarios with 3G path(s), a slight 1273 1237 benefit can be seen: the 3G path has a small ca- 1274 1238 pacity and also a higher latency. Therefore, com- 1275 1239 bining this 3G path with another 3G path, or even 1276 1240 with a WLAN path, results in a faster download of 1277 1241

the Wikipedia web site. As expected, for the Amazon (1 MiB) and the Huffington Post (3.9 MiB) web sites, CMT-SCTP reaches a significant download time reduction in most cases. However, for the WLAN-3G scenario with background traffic the path asymmetry is too large and CMT-SCTP performs worse than SCTP over WLAN for all web sites. Here the negative effect from head-of-line blocking dominates the gain from load balancing.

### 5.3.2. MPTCP Emulation

The average web site download times over 30 runs for the three web site scenarios, without background traffic, are presented in Figure 18. The corresponding results with background traffic are presented in Figure 19. Comparing the delay performance of MPTCP to that of TCP, MPTCP only provides limited improvements in download time for Wikipedia, but larger gains for both Amazon and Huffington Post (especially in 3G-3G scenarios). As also seen for CMT-SCTP, the results indicate that the size of the web site is critical to the total download time. With concurrent connections (6 in our setup), small web sites such as Wikipedia, can mostly be transferred within the initial window of TCP, not allowing MPTCP to exploit multiple paths.

With background traffic, the performance trends are similar to that of the non-background case. For the 3G paths, background traffic significantly increases the download time as well as the variation in download time. Background traffic has very little impact over the WLAN paths, but the random loss over the WLAN links still leads to a large variation in download times. The position of the loss in the short web flows can have a huge impact on the total web site download duration.



Figure 13: Average Message Delay for Gaming Traffic over MPTCP in NorNet Experiment



Figure 14: Website Download Times over CMT-SCTP (without Background Traffic)







(a) Wikipedia (www.wikipedia.org)

(b) Amazon (www.amazon.com)

(c) Huffpost (www.huffingtonpost.com)

Figure 15: Website Download Times over CMT-SCTP (with Competing Background Traffic)

Website	Background	WLAN-WLAN		3G-3G		WLAN-3G	
	Dackground	WLAN	WLAN	3G	$3\mathrm{G}$	WLAN	3G
Wikipedia	×	23.5	76.5	21.7	78.3	24.5	75.5
	1	23.7	76.3	23.3	76.7	33.0	67.0
Amazon	X	24.6	75.4	23.6	76.4	31.2	68.8
Amazon	1	24.3	75.7	24.1	75.9	44.2	54.8
Huffington Post	×	23.9	76.1	23.2	76.8	31.9	68.1
	1	23.9	76.1	24.6	76.4	50.5	49.5

Table 10: Path 1 Traffic Share (in %) for Website Download over CMT-SCTP



Figure 16: Website Download Times over MPTCP in CORE Emulation (without Background Traffic)



Figure 17: Website Download Times over MPTCP in CORE Emulation (with Competing Background Traffic)



Figure 18: Website Download Times over MPTCP in NorNet Experiment

Table 11 indicates that in symmetric scenarios, 1326 1278 data transfer uses both paths. In WLAN-WLAN, 1327 1279 random losses causes data to be sent over the sec- 1328 1280 ond path even in small web sites like Wikipedia. 1329 1281 The delay variance on the paths is the primary rea- 1330 1282 son for use of the second path in 3G-3G scenarios. 1331 1283 In asymmetric scenarios, most of the data uses the 1332 128 primary faster path WLAN. 1333 1285

### 1286 5.3.3. MPTCP Real Measurements

In Figure 20, we illustrated the results of down-  $^{1336}$ 1287 load times for web traffic. For WLAN-WLAN, we <sup>1337</sup> 1288 observe that the MPTCP delay is lower than the <sup>1338</sup> 1289 average TCP delay for Amazon whereas it is higher <sup>1339</sup> 1290 than the average TCP delay for the other two web  $^{\scriptscriptstyle 1340}$ 1291 sites: Wikipedia and Huffington Post. Similar ob-<sup>1341</sup> 1292 servations can be made for the 3G-3G scenario.  $^{\scriptscriptstyle 1342}$ 1293 We observe that when there is enough data to be  $^{1343}$ 1294 transmitted, MPTCP can provide benefits. For the <sup>1344</sup> 1295 WLAN-3G case, similar to other traffic, we observe <sup>1345</sup> 1296 that for all the web sites, MPTCP delay is a little  $^{\rm 1346}$ 1297 bit higher than the WLAN delay. 1298

The traffic distribution of data over the different <sup>1348</sup> paths is shown in Table 12. We observed that for <sup>1349</sup> the heterogeneous cases, almost all traffic is trans- <sup>1350</sup> ferred over the WLAN path whereas for the homo-<sup>1351</sup> geneous cases, the distribution depends on the size <sup>1352</sup> of the web site. <sup>1353</sup>

### 1305 6. Discussion of Results

1357

1354

1355

1356

1334

1335

In this paper, we have run extensive measure- 1358 1306 ments to evaluate the capability of multi-path 1359 1307 transport protocols to carry latency sensitive ap- 1360 1308 plication traffic. More specifically, we have anal-1361 1309 ysed the application delays for video traffic, online 1362 1310 gaming and web services both with and without 1363 1311 competing traffic. Furthermore, we considered end- 1364 1312 hosts experiencing multiple homogeneous paths as 1365 1313 well as heterogeneous ones. The results are summa- 1366 1314 rized in Figure 21, where the performance of multi- 1367 1315 path, as compared to single path, is categorized into 1368 1316 four types. We next elaborate on this table and re- 1369 1317 cap our findings from Section 5. 1318 1370

The currently used round-robin scheduler in 1371 CMT-SCTP is optimised for throughput, not for 1372 low latency. Therefore, when there is a significant 1373 delay difference between the two paths, we observe 1374 performance degradation with CMT-SCTP com- 1375 pared to SCTP. However, for homogeneous paths, 1376 especially when the paths are not very lossy, we 1377 observe that CMT-SCTP can significantly reduce latency, especially in the web scenario. For video traffic CMT-SCTP provides similar or higher delay values as compared to SCTP. For example, in WLAN-WLAN scenarios, reordering due to distribution over multiple lossy paths increases the delay both with and without background traffic. On the other hand, in 3G-3G scenarios, we observe similar delay values for CMT-SCTP and SCTP. For the WLAN-3G scenario, packet reordering causes increased delay for CMT-SCTP even when there is no competing traffic. For gaming traffic, while CMT-SCTP leads to a latency increase in case of two similar, low-latency WLAN paths, it becomes beneficial in case of high-latency paths with background traffic; in comparison of using only the higher-delay path, CMT-SCTP is able to take advantage of the lower-delay path to reduce latency. However, as observed for the video traffic, its scheduler is optimised for throughput maximization without taking care of path delay. Therefore, while the latency is lower than using only the high-delay path, it is still much higher than using only the low-delay path in the heterogeneous WLAN-3G case. For web traffic, we observe that using CMT-SCTP improves the web site download speed, especially for homogeneous paths. Since the web traffic is saturated (i.e., send as much data as possible), the roundrobin scheduler that is used by the CMT-SCTP implementation performs reasonably well by ensuring that both paths are utilized, although it does not always chose the path with the lowest RTT. The larger the web site, the better the performance improvement achieved by CMT-SCTP. However, for the WLAN-3G case, especially when there is background traffic, CMT-SCTP cannot handle the delay difference between the 3G and WLAN paths, resulting in poor performance for CMT-SCTP compared to SCTP.

For MPTCP, the default scheduler is based on delay and it has been shown to achieve low and stable latency [?]. For different type of traffics, we observe similar or lower delay values for MPTCP compared to TCP, for homogeneous paths. However, the main factor that determines the delay performance of MPTCP is indeed the path heterogeneity, and for heterogeneous paths we observe performance degradation whose degree depends on the traffic and whether there exists background traffic. More specifically, for video traffic, when the links are lossy, e.g. the WLAN-WLAN case, we observe delay gains due to link aggregation. For the 3G-3G

	Background	WLAN-	WLAN-WLAN		3G-3G		-3G
	Dackground	WLAN	WLAN	3G	3G	WLAN	3G
Wikipodia	×	90.22	9.78	86.98	13.02	99.94	0.06
wikipeula	1	86.98	13.02	83.86	16.14	99.94	0.06
Amazon	×	76.02	23.98	56.73	43.27	99.81	0.19
Amazon	1	70.45	29.55	72.64	27.36	99.9	0.1
Huffington Post	×	85.3	14.7	68.1	31.9	99.9	0.1
	1	83.0	17	79.13	20.87	98.86	1.14

Table 11: Web traffic data share per path using MPTCP

	WLAN-	-WLAN	3G-3	3G	WLAN-3G	
	WLAN	WLAN	3G	3G	WLAN	3G
Wikipedia	94.3	5.6	100.0	0.0	100.0	0.0
Amazon	26.1	73.8	96.9	3.1	100.0	0.0
Huffington Post	14.5	85.4	100.0	0.0	100.0	0.0

Table 12: Web traffic data share per path using MPTCP in NorNet

	МРТСР			CMT-SCTP		
Traffic	Symmetric		Asymmetric	Symmetric		Asymmetric
	WLAN-WLAN	3G - 3G	WLAN - 3G	WLAN-WLAN	3G - 3G	WLAN - 3G
Video						
Video with BG						
Gaming						
Gaming with BG						
Wikipedia						
Wikipedia with BG						
Amazon						
Amazon with BG						
Huffpost						
Huffpost with BG						
Significant performance improvement with multi-path than that of single path Slight performance improvement with multi-path than that of single path						

Slight performance improvement with multi-path than that of single path No improvement with multi-path but no significant degradation Performance degraded significantly with multi-path

Figure 19: Multi-path versus single path transport protocols depending on the latency sensitive traffic: Summary table

case, where there are no losses, MPTCP selects the 1430 1378 best available path resulting in minor gains com-1431 1379 pared to TCP. In the presence of background traffic, 1432 1380 the delay values are in general higher but the ben- 1433 1381 efits of using MPTCP are consistent with that of 1434 1382 the non-competing traffic. We further observe that 1435 1383 these emulation results are mostly consistent with 138 the NorNet experiments. Here, the main difference 1385 is that in real networks, paths have more diverse 1436 1386 characteristics, although using the same technology. 1387 This results in slight heterogeneity and the delay us- 1437 1388 ing MPTCP becomes higher than using TCP over 1438 1389 the best path only, but still lower than the average 1439 1390 TCP delay. For gaming traffic, we observe very sim- 1440 1391 ilar delay values to TCP for almost all cases due to 1441 1392 the very limited amount of data. The background 1442 1393 traffic did not induce enough loss in the foreground 1443 1394 flows, therefore the delay values are similar to that 1444 1395 of no background traffic. One exception is the 3G-1445 1396 3G scenario where MPTCP keeps sending over one 1446 1397 path as long as there is no loss, therefore, pro-1447 1398 viding some delay gains. For the WLAN-3G case, 1448 1399 MPTCP uses the WLAN at almost all times, there- 1449 1400 fore, the MPTCP delay is similar to the TCP delay 1450 1401 of WLAN. Similarly, for the results of the real ex- 1451 1402 periments, we observe similar behavior to the emu- 1452 1403 lations, especially for the homogeneous scenarios, 1453 1404 with slight variations among different trace files. 1454 1405 For the heterogeneous scenario, there is slightly 1455 1406 higher delay in real experiments compared to emu-1456 1407 lations, due to some traffic is being transferred over 1457 1408 the slower 3G path. For the web traffic, we observe 1458 1409 that MPTCP provides lower delay values, especially 1459 1410 for web sites with many objects. The lower delay is 1460 1411 a consequence of MPTCP's scheduler which always 1461 1412 tries to use the path with the lowest RTT. However, 1462 1413 when the paths are very heterogeneous in terms of  $_{\rm 1463}$ 141 delay and loss as in the WLAN-3G case, losses in 1464 1415 WLAN forces MPTCP to use the suboptimal 3G 1465 1416 path; therefore, the MPTCP delay becomes higher 1466 1417 than the TCP delay of WLAN. These results hold 1467 1418 for emulations with and without background traf-1468 1419 fic. Similar to the previous applications, the results 1469 1420 of the real experiments are mostly consistent with 1470 1421 the emulation results. Due to the differences in the 1471 1422 paths for the homogeneous cases (e.g. 3G-3G and 1472 1423 WLAN-WLAN), MPTCP delay is higher than the 1473 1424 best path while still much lower than the average 1474 1425 TCP delay. 1426 1475

<sup>1427</sup> One conclusion of our study is that multi-path <sup>1476</sup> <sup>1428</sup> transport protocols can hardly reduce the latency <sup>1477</sup> <sup>1429</sup> for all the tested applications, when there is some <sup>1478</sup> asymmetry between the paths. Moreover, in this case, multi-path transport may increase the latency, mainly because of head-of-line blocking. However, it is worth pointing out that in most symmetric scenarios, multi-path transport protocols enable a significant latency reduction.

### 7. Related Work

This section discusses work related to ours. While there are numerous articles on MPTCP and its performance in relation to TCP, not much has been written on the relation between CMT-SCTP and SCTP. Instead, most articles on CMT-SCTP propose various optimizations to the protocol itself. There are, of course exceptions; Aydin et al. [?] elaborates the importance of TCP friendliness for single homed SCTP and evaluates the TCP friendliness of CMT-SCTP. Arianpoo et al. [?] propose an adaptive network coding mechanism for CMT-SCTP to desensitize the receiver against packet reordering and consequently eliminate the receiver buffer blocking problem. They claim to have improve the CMT-SCTP performance by 62% over the original implementation in cases of severe path asymmetry.

For MPTCP, a closely related research work is [62], which measures MPTCP performance with the aim to understand the benefit of using two interfaces with MPTCP over using either one of the interface with TCP. This study also focuses on the impact that flow size has on the average latency, and provides insights into the effect of path characteristic diversity on application level performance. Their conclusions are consistent with ours: using multi-path becomes more and more beneficial when the size of the data to transmit increases. We extend their work by considering more application scenarios. In [63], S. Deng et al., studies the performance of MPTCP over wireless technologies using Android application traffic. Their study focuses on energy efficiency and provides new challenges such as dynamic decision making at the mobile applications to select appropriate network technology depending on the flow size and traffic pattern. Handover performance was seen as a potential MPTCP performance impairment especially when the path characteristics are different. Andrei et al. [64] provided a simultaneous association solution using MPTCP for WLAN that avoids fast handover. It also provides possible modifications at the

client side implementation, to mitigate the through-1531 1479 put loss in cases where the WLAN characteristics 1532 1480 differ due to channel specification. Such approach 1533 1481 of reducing the occurrence of handovers is a nec- 1534 1482 essary improvement: it is essential to improve per- 1535 1483 formance where multi-path transport increases the 1536 1484 latency, and that has been identified in this paper. 1537 1485 Grinnemo et al. [65] provides a first comprehen- 1538 1486 sive evaluation of MPTCP performance with la- 1539 1487 tency as the quality of experience metric for cloud- 1540 1488 based applications. They study three different ap- 1541 1489 plications: Netflix, Google Maps and Google Docs, 1490

representing high, mid and low intensity cloud-1491 1542 based traffic. The authors conclude that MPTCP 1492 provides significant performance gains for high and 1543 1493 mid intensity traffic. Furthermore, it is noted that 1544 1494 the variation in RTTs among network paths causes 1545 1495 higher application latency, and the current Linux 1546 1496 standard scheduler is seen as the primary cause of 1547 1497 increased latency in such cases. 1548 1498

Raiciu et al. [60] proposed a mobility architec- 1549 1499 ture to allow MPTCP to switch between differ- 1550 1500 ent technologies and handle mobility at the trans- 1551 1501 port layer instead of at the network layer. The 1552 1502 mobility of MPTCP was evaluated with simula- 1553 1503 tions and indoor mobility experiments. The cri- 1554 1504 teria for the evaluation was measured throughput 1555 1505 on TCP and MPTCP using WLAN-3G, and power 1556 1506 efficiency of both protocols. The study concludes 1557 1507 that MPTCP provides performance improvements 1558 1508 over TCP when multiple interfaces are used in par- 1559 1509 allel. Power efficiency of MPTCP depends on the 1560 1510 underlying interface power consumption and should 1561 1511 be tuned for better performance. Later, the power 1562 1512 efficiency of MPTCP drew much attention in [61], 1563 1513 which analyses the energy consumption and han-1564 1514 dover performance of MPTCP in the different oper- 1565 1515 ational modes: Full MPTCP Mode, Backup Mode 1566 1516 and Single path Mode. This work again provides 1567 1517 experimental evaluations using the Linux imple-1568 1518 mentation of MPTCP and commercial access net- 1569 1519 works providing 3G and broadband access on static 1570 1520 The study concludes that MPTCP han- 1571 nodes. 1521 dovers might have small impact on application de- 1572 1522 lay and goodput in different operational modes. 1573 1523

With a few exceptions discussed above, most of 1574
the prior research on CMT-SCTP or MPTCP mea- 1575
surements focused on the performance of the pro-

tocol in terms of throughput, energy consumption,
handover performance and RTTs. To the best of
our knowledge, this paper is the first to provide 1577
a comprehensive analysis of multi-path transport 1578

performance with latency as the main metric.

As seen throughout this paper, the performance of multi-path transport is highly dependent on doing efficient scheduling. Paasch et al. [24] provide a detailed study of schedulers and their impact on performance. The authors implement a generic modular framework for evaluating MPTCP schedulers in Linux. Using this framework, different schedulers are then evaluated using various performance metrics and different types of traffic, including bulk and application limited traffic.

### 8. Conclusions and Future Work

For an increasing number of applications, latency plays an important role as it directly impacts their performance. Still, most work considering multipath communication is solely focused on resilience and throughput maximization. The work presented in this paper tries to bridge this gap by evaluating whether multi-path communication can help latency-sensitive applications satisfy their users' requirements. Three latency-sensitive applications have been considered: video, gaming and web traffic. Performance have been evaluated using 3G– 3G, 3G–WLAN, and WLAN–WLAN paths, in both simulated, emulated and real-life environments considering both CMT-SCTP and MPTCP.

The results indicate that multi-path communication can reduce latency significantly, but only when paths are symmetric in terms of delay and loss rate. The potential gain comes mainly from two factors: the possibility to distribute short bursts of data over multiple interfaces and the ability to select the best of the available paths for data transmission. In asymmetric scenarios where the latency reduction is not as significant (or non-existent), applications may still benefit from other properties of multi-path communication, without increasing latency. This is, however, highly dependent on the scheduling mechanism used. As seen in some of the CMT-SCTP experiments, a scheduler designed mainly for throughput maximization, may lead to increased latency in some scenarios. Considering the importance of scheduling, this is where we direct our attention for future work, and we are currently designing a scheduler targeting latency-sensitive traffic.

### Acknowledgments

The authors are funded by the European Community under its Seventh Framework Programme

through the Reducing Internet Transport La- 1640 1579 tency (RITE) project (ICT-317700). The views ex-<sup>1641</sup> 1580 1642

1643

1644

1645

1646

1647

pressed are solely those of the authors. 1581

#### References 1582

1602

1604

1605

1606

1607

1614

1615

1616

1617

1618

1619

1620

1629

1631

- "Video <sup>1648</sup> [1] S. S. Krishnan and R. K. Sitaraman, 1583 1649 Stream Quality Impacts Viewer Behavior: Inferring 1584 1650 Causality Using Quasi-experimental Designs," in 1585 Proceedings of the 12th ACM Internet Measurement 1651 1586 1652 Conference (IMC). New York/U.S.A.: ACM, 2012, 1587 1653 pp. 211-224. [Online]. Available: http://doi.acm.org/ 1588 1654 10.1145/2398776.2398799 1589 1655
- P. Quax, P. Monsieurs, W. Lamotte, D. D. [2]1590 Vleeschauwer, and N. Degrande, "Objective and <sup>1656</sup> 1591 1657 Subjective Evaluation of the Influence of Small 1592 Amounts of Delay and Jitter on a Recent First Person Shooter Game," in *Proceedings of 3rd* 1658 1593 3rd <sup>1659</sup> 1594 1660 ACM SIGCOMM Workshop on Network and Sys-1595 tem Support for Games, ser. NetGames '04. New <sup>1661</sup> 1596 1662 York/U.S.A.: ACM, 2004, pp. 152–156. [Online]. Avail-1597 1663 able: http://doi.acm.org/10.1145/1016540.1016557 1598 1664
- [3] A. Ford, C. Raiciu, M. Handley, and O. Bonaventure, 1599 1665 "TCP Extensions for Multipath Operation with 1600 1666 Multiple Addresses," IETF, RFC 6824, Jan. 2013, 1601 1667 ISSN 2070-1721. [Online]. Available: https://tools.ietf. 1668 org/rfc/rfc6824.txt 1603 1669
  - J. B. Postel, "Transmission Control Protocol," IETF, [4]1670 Standards Track RFC 793, Sep. 1981, ISSN 2070-1721. 1671 [Online]. Available: https://www.ietf.org/rfc/rfc793. 1672 tvt
- 1673 [5] J. R. Iyengar, P. D. Amer, and R. R. Stewart, "Concur-1608 rent Multipath Transfer using SCTP Multihoming over <sup>1674</sup> 1609 Independent End-to-End Paths," IEEE/ACM Trans- 1675 1610 1676 actions on Networking, vol. 14, no. 5, pp. 951-964, 1611 1677 Oct. 2006, ISSN 1063-6692. [Online]. Available: http: 1612 1678 www.fandm.edu/jiyengar/papers/cmt-ton2006.pdf 1613 1679
  - P. D. Amer, M. Becke, T. Dreibholz, N. Ekiz, J. R. [6] 1680 Iyengar, P. Natarajan, R. R. Stewart, and M. Tüxen, 1681 "Load Sharing for the Stream Control Transmission 1682 Protocol (SCTP)," IETF, Individual Submission, 1683 Internet Draft draft-tuexen-tsvwg-sctp-multipath-10, 1684 May 2015. [Online]. Available: https://tools.ietf.org/ 1685 id/draft-tuexen-tsvwg-sctp-multipath-10.txt
- "Evaluation and Optimisation of <sup>1686</sup> Dreibholz, [7]Т. 1621 Multi-Path Transport using the Stream Control 1687 1622 1688 Protocol," Treatise, Transmission Habilitation 1623 1689 University of Duisburg-Essen, Faculty of Economics, 1624 1690 Institute for Computer Science and Business 1625 Information Systems, Mar. 2012. [Online]. Available: <sup>1691</sup> 1626 https://duepublico.uni-duisburg-essen.de/servlets/ 1627 1693 1628 DerivateServlet/Derivate-29737/Dre2012\_final.pdf 1694
- R. R. Stewart, "Stream Control Transmission Proto-1695 col," IETF, RFC 4960, Sep. 2007, ISSN 2070-1721. [On-1630 1696 line]. Available: https://tools.ietf.org/rfc/rfc4960.txt
- 1697 IETF SIGTRAN WG, "IETF Signaling Transport 1632 1698 Working Group," 2012. [Online]. Available: https:// 1633 1699 1634 /datatracker.ietf.org/wg/sigtran/charter/
- 1700 ITU-T, "Introduction to CCITT Signalling Sys-[10]1635 tem No. 7," International Telecommunication Union, 1701 1636 Recommendation Q.700, May 1994. [Online]. Avail-<sup>1702</sup> 1637 1703 able: https://www.itu.int/rec/dologin\_pub.asp?lang= 1638 1704 e&id=T-REC-Q.700-199303-I!!PDF-E&type=items 1639

- [11] J. B. Postel, "Internet Protocol," IETF, Standards Track RFC 791, Sep. 1981, ISSN 2070-1721. [Online]. Available: https://www.ietf.org/rfc/rfc791.txt
- -, "User Datagram Protocol," IETF, Standards [12]Track RFC 768, Aug. 1980, ISSN 2070-1721. [Online]. Available: https://www.ietf.org/rfc/rfc768.txt
- R. R. Stewart, Q. Xie, M. Tüxen, S. Maruyama, [13]and M. Kozuka, "Stream Control Transmission Protocol (SCTP) Dynamic Address Reconfiguration," IETF, Standards Track RFC 5061, Sep. 2007, ISSN 2070-1721. [Online]. Available: https://tools.ietf.org/ rfc/rfc5061.txt
- [14] L. Budzisz, J. Garcia, A. Brunström, and R. Ferrús, "A Taxonomy and Survey of SCTP Research," ACM Computing Surveys (CSUR), vol. 44, no. 4, pp. 18:1–18:36, Sep. 2012. [Online]. Available: http: //doi.acm.org/10.1145/2333112.2333113
- [15] M. Allman, V. Paxson, and E. Blanton, "TCP Congestion Control," IETF, Standards Track RFC 5681, Sep. 2009, ISSN 2070-1721. [Online]. Available: https://tools.ietf.org/rfc/rfc5681.txt
- [16] S. Barré, C. Paasch, and O. Bonaventure, "MultiPath TCP: From Theory to Practice," in Proceedings of the 10th International IFIP Networking Conference, Valencia/Spain, May 2011, pp. 444-457, ISBN 978-3-642-20756-3. [Online]. Available: http://inl.info.ucl.ac. be/system/files/networking-mptcp.pdf
- [17] IETF MPTCP WG, "IETF Multipath TCP Working Group," 2012. [Online]. Available: https://datatracker. ietf.org/wg/mptcp/charter/
- C. Raiciu, M. Handley, and D. Wischik, "Coupled Con-[18]gestion Control for Multipath Transport Protocols," IETF, RFC 6356, Oct. 2011, ISSN 2070-1721. [Online]. Available: https://tools.ietf.org/rfc/rfc6356.txt
- [19] C. Raiciu, S. Barré, C. Pluntke, A. Greenhalgh, D. Wischik, and M. Handley, "Improving Datacenter Performance and Robustness with Multipath TCP," in Proceedings of the ACM SIGCOMM Conference, Toronto, Ontario/Canada, Aug. 2011, pp. 266-277, ISBN 978-1-4503-0797-0. [Online]. Available: http: //inl.info.ucl.ac.be/system/files/mptcp-sigcomm.pdf
- [20] L. Boccassi, M. Fayed, and M. Marina, "Binder: a system to aggregate multiple internet gateways in community networks," in Proceedings of the ACM Mobi-Com Workshop on Lowest Cost Denominator Networking for Universal Access (LCDNet), Miami, FL, USA, Sep. 2013, pp. 3-8.
- [21] H. Adhari, T. Dreibholz, M. Becke, E. P. Rathgeb, and M. Tüxen, "Evaluation of Concurrent Multipath Transfer over Dissimilar Paths," in Proceedings of the 1st International Workshop on Protocols and Applications with Multi-Homing Support (PAMS), Singapore, Mar. 2011, pp. 708-714, ISBN 978-0-7695-4338-3. [Online]. Available: https://www.wiwi.uni-due.de/fileadmin/ fileupload/I-TDR/SCTP/Paper/PAMS2011.pdf
- [22]T. Dreibholz, M. Becke, E. P. Rathgeb, and M. Tüxen, "On the Use of Concurrent Multipath Transfer over Asymmetric Paths," in Proceedings of the IEEE Global Communications Conference (GLOBECOM), Miami, Florida/U.S.A., Dec. 2010,ISBN 978-1-4244-5637-6. [Online]. Available: https://www.wiwi.uni-due.de/fileadmin/ fileupload/I-TDR/SCTP/Paper/Globecom 2010.pdf
- [23]N. Kuhn, E. Lochin, A. Mifdaoui, G. Sarwar, O. Mehani, and R. Boreli, "DAPS: Intel-

ligent Delay-Aware Packet Scheduling For Multi-1770 path Transport," in *IEEE International Conference* 1771 on Communications (ICC), Sydney, New South 1772 Wales/Australia, Jun. 2014. [Online]. Available: http://oatao.univ-toulouse.fr/11767/1/Lochin\_11767.pdf 1774

- 1709 C. Paasch, S. Ferlin, Özgü Alay, and O. Bonaven- 1775 [24]1710 ture, "Experimental Evaluation of Multipath TCP 1776 1711 Schedulers," in Proceedings of the ACM SIG-1777 1712 COMM Capacity Sharing Workshop (CSWS), 1778 1713 Illinois/U.S.A., Aug. 2014,pp. 27– 1779 Chicago, 1714 32.ISBN 978-1-4503-2991-0. [Online]. Avail- 1780 1715 https://web.archive.org/web/20141127063815/ 1781 1716 able: http://inl.info.ucl.ac.be/system/files/paper\_7.pdf 1717 1782
- [25]C. Raiciu, C. Paasch, S. Barré, A. Ford, M. Honda, 1783 1718 F. Duchêne, O. Bonaventure, and M. Handley, 1784 1719 "How Hard Can It Be? Designing and Implement- 1785 1720 ing a Deployable Multipath TCP," in Proceedings 1766 of the 9th USENIX Conference on Networked 1787 1721 1722 Systems Design and Implementation (NSDI), San 1788 1723 Jose, California/U.S.A., Apr. 2012, pp. 1–14. [On- 1789 1724 line]. Available: https://www.usenix.org/system/files/ 1790 1725 conference/nsdi12/nsdi12-final125.pdf 1791 1726
- [26] T. Dreibholz, M. Becke, J. Pulinthanath, and E. P. 1792 1727 1728 Rathgeb, "Applying TCP-Friendly Congestion Control 1793 to Concurrent Multipath Transfer," in Proceedings of 1794 1729 the 24th IEEE International Conference on Advanced 1795 1730 Information Networking and Applications (AINA), 1796 1731 Perth, Western Australia/Australia, Apr. 2010, 1797 1732 pp. 312-319, ISBN 978-0-7695-4018-4. [Online]. 1798 1733 Available: https://www.wiwi.uni-due.de/fileadmin/ 1799 1734 fileupload/I-TDR/SCTP/Paper/AINA2010.pdf 1735 1800
- T. Dreibholz, M. Becke, H. Adhari, and E. P. 1801 1736 [27]Rathgeb, "On the Impact of Congestion Con- 1802 1737 trol for Concurrent Multipath Transfer on the 1803 1738 Transport Layer," in Proceedings of the 11th 1804 1739 IEEE International Conference on Telecommunica- 1805 1740 1741 tions (ConTEL), Graz, Steiermark/Austria, Jun. 1806 2011, pp. 397-404, ISBN 978-953-184-152-8. [Online]. 1807 1742 https://www.wiwi.uni-due.de/fileadmin/ 1808 1743 Available: fileupload/I-TDR/SCTP/Paper/ConTEL2011.pdf 1744 1809
- R. Khalili, N. Gast, M. Popović, and J.-Y. L. 1810 [28]1745 1746 Boudec, "MPTCP is not Pareto-Optimal: Perfor- 1811 1747 mance Issues and a Possible Solution," IEEE/ACM 1812 Transactions on Networking, vol. 21, no. 5, pp. 1813 1748 1651-1665, Oct. 2013, ISSN 1063-6692. [Online]. Avail- 1814 1749 able: http://conferences.sigcomm.org/co-next/2012/ 1815 1750 eproceedings/conext/p1.pdf 1751 1816
- [29] A. Walid, Q. Peng, J. Hwang, and S. H. Low, 1817
  "Balanced Linked Adaptation Congestion Control 1818
  Algorithm for MPTCP," IETF, Individual Submission, 1819
  Internet Draft draft-walid-mptcp-congestion-control- 1820
  02, Jan. 2015. [Online]. Available: https://tools.ietf. 1821
  org/id/draft-walid-mptcp-congestion-control-02.txt
  1822
- [30] J. R. Iyengar, P. D. Amer, and R. R. Stewart, "Re-1823 transmission policies for Concurrent Multipath Transfer 1824 using SCTP Multihoming," in *Proceedings of the 12th* 1825 *IEEE International Conference on Networks (ICON)*, 1826 vol. 2, 2004, pp. 713–719.
- [31] C. Labovitz, S. Iekel-Johnson, D. McPherson, 1828 1763 J. Oberheide, F. Jahanian, and M. Karir, "Atlas 1829 1764 Internet Observatory 2009 Annual Report," in 1830 1765 47th NANOG, 2009. [Online]. Available: http: 1831 1766 //www.nanog.org/meetings/nanog47/presentations/ 1767 1832 1768 Monday/Labovitz\_ObserveReport\_N47\_Mon.pdf 1833
- 1769 [32] D. Ciullo, M. Mellia, and M. Meo, "Two 1834

Schemes to Reduce Latency in Short Lived TCP Flows," *IEEE Communications Letters*, vol. 13, no. 10, pp. 806–808, 2009. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/download? doi=10.1.1.159.1967&rep=rep1&type=pdf

- [33] Sandvine Intelligent Broadband Networks, "Global Internet Phenomena Report," Jul. 2013.
   [Online]. Available: https://web.archive.org/ web/20141216103806/https://www.sandvine.com/ downloads/general/global-internet-phenomena/2013/ sandvine-global-internet-phenomena-report-1h-2013.
   pdf
- [34] Cisco Visual Networking Index, "Forecast and Methodology, 2012–2017," 2013. [Online]. Available: https://web.archive.org/web/20141216103933/http: //www.cisco.com/c/en/us/solutions/collateral/ service-provider/ip-ngn-ip-next-generation-network/ white\_paper\_c11-481360.pdf
- [35] X. Zhang, Y. Xu, H. Hu, Y. Liu, Z. Guo, and Y. Wang, "Profiling Skype Video Calls: Rate Control and Video Quality," in *Proceedings of the IEEE INFOCOM*, Orlando, Florida/U.S.A., Mar. 2012, pp. 621–629. [Online]. Available: http://eeweb.poly.edu/ faculty/yongliu/docs/skype\_congestion\_infocom.pdf
- [36] L. D. Cicco, S. Mascolo, and V. Palmisano, "Skype Video Congestion Control: An Experimental Investigation," *Computer Networks*, vol. 55, no. 3, pp. 558–571, 2011. [Online]. Available: http://www.sciencedirect. com/science/article/pii/S1389128610002914
- [37] T. Szigeti and C. Hattingh, End-to-End QoS Network Design: Quality of Service in LANs, WANs, and VPNs Quality of Service Design Overview. Cisco Press, Nov. 2004, ISBN 978-1587051760.
- [38] M. Claypool and K. Claypool, "Latency and Player Actions in Online Games," Communications of the ACM, vol. 49, no. 11, pp. 40–45, Nov. 2006. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/ download?doi=10.1.1.216.9904&rep=rep1&type=pdf
- [39] X. Che and B. Ip, "Review: Packet-level Traffic Analysis of Online Games from the Genre Characteristics Perspective," Journal of Network and Computer Applications, vol. 35, no. 1, pp. 240–252, Jan. 2012. [Online]. Available: http://dx.doi.org/10.1016/j.jnca.2011.08.005
- [40] N. Sheldon, E. Girard, S. Borg, M. Claypool, and E. Agu, "The effect of latency on user performance in Warcraft III," in *Proceedings of the* 2nd Workshop on Network and System Support for Games (NetGames), Redwood City, California/U.S.A., May 2003, pp. 3–14. [Online]. Available: http: //web.cs.wpi.edu/~emmanuel/TR-03-07.pdf
- [41] K.-T. Chen, P. Huang, G.-S. Wang, C.-Y. Huang, and C.-L. Lei, "On the Sensitivity of Online Game Playing Time to Network QoS," in *Proceedings of the IEEE IN-FOCOM*, Barcelona, Catalonia/Spain, Apr. 2006, pp. 1–12.
- [42] O3b Networks and Sofrecom, "Why Latency Matters to Mobile Backhaul," Apr. 2013. [Online]. Available: https://web.archive.org/web/20141216102926/ http://www.o3bnetworks.com/media/45606/ o3b\_latency\_mobile%20backhaul\_130417.pdf
- [43] Google Developers, "Web Metrics: Size and Number of Resources," May 2010. [Online]. Available: https://web.archive.org/web/20140625155405/https: //developers.google.com/speed/articles/web-metrics

- 1835 [44] A. Varga, OMNeT++ Discrete Event Simulation 1900
  1836 System User Manual Version 4.6, Dec. 2014. [On- 1901
  1837 line]. Available: https://omnetpp.org/doc/omnetpp/ 1902
  1838 Manual.pdf 1903
- 1839 [45]T. Dreibholz, M. Becke, J. Pulinthanath, and 1904 P. Rathgeb, "Implementation and Evaluation 1905 E. 1840 of Concurrent Multipath Transfer for SCTP in 1906 1841 the INET Framework," in Proceedings of the 3rd 1907 1842 ACM/ICST International Workshop on OMNeT++, 1908 1843 Torremolinos, Málaga/Spain, Mar. 2010, ISBN 1909 1844 978-963-9799-87-5. [Online]. Available: https: 1910 1845 //www.wiwi.uni-due.de/fileadmin/fileupload/I-TDR/ 1846 1911 SCTP/Paper/OMNeT\_\_Workshop2010-SCTP.pdf 1847 1912
- I. Rüngeler, "SCTP Evaluating, Improving and 1913 [46]1848 Extending the Protocol for Broader Deployment," 1849 1914 Ph.D. dissertation, University of Duisburg-Essen, 1915 1850 Faculty of Economics, Institute for Computer Science 1916 1851 1852 and Business Information Systems, Dec. 2009. [Online]. 1917 Available: https://duepublico.uni-duisburg-essen.de/ 1918 1853 servlets/DerivateServlet/Derivate-36393/Diss.pdf 1854 1919
- P. 1920 T. Dreibholz, M. Becke, H. Adhari, and E. 1855 [47]Rathgeb, "Evaluation of A New Multipath Congestion 1921 1856 Control Scheme using the NetPerfMeter Tool-Chain," 1857 1922 1858 in Proceedings of the 19th IEEE International 1923 Conference on Software, Telecommunications and 1924 1859 Computer Networks (SoftCOM), Hvar/Croatia, Sep. 1925 1860 2011, pp. 1-6, ISBN 978-953-290-027-9. [Online]. 1926 1861 https://www.wiwi.uni-due.de/fileadmin/ 1927 1862 Available: fileupload/I-TDR/SCTP/Paper/SoftCOM2011.pdf 1863 1928
- 1864[48] A. Varga, INET Framework for OMNeT++, 2015.1865[Online]. Available: https://inet.omnetpp.org/
- [49] J. Ahrenholz, "Comparison of CORE Network Emulation Platforms," in *Military Communications Conison ference (MILCOM)*, San Jose, California/U.S.A., Oct. 1933
  2010, pp. 166–171.
- [50]A. Kvalbein, D. Baltrūnas, K. R. Evensen, J. Xi- 1935 1870 "The 1936 1871 ang, A. M. Elmokashfi, and S. Ferlin, NorNet Edge Platform for Mobile Broadband 1937 1872 Measurements." Computer Networks, Special Issue 1938 1873 on Future Internet Testbeds, vol. 61, pp. 88-1939 1874 101, Mar. 2014, ISSN 1389-1286. [Online]. Avail- 1940 1875 able: https://www.simula.no/sites/www.simula.no/ 1941 1876 files/publications/Simula.simula.2434.pdf1942 1877
- Dreibholz, S. Ferlin, O. Alay, N. Kuhn, 1943 [51]Т. 1878 Κ. Yedugundla, P. Hurtig, and A. Brunstrom, 1944 1879 "Is multi-path transport suitable for latency sen- 1945 1880 sitive traffic," Sep. 2015. [Online]. Available: 1946 1881 https://git.cs.kau.se/rite/multipath-latency 1882 1947
- 1883
   [52] M. Allman and E. Blanton, "Notes on Burst Mit- 1948

   1884
   igation for Transport Protocols," ACM SIGCOMM 1949

   1885
   Computer Communication Review (CCR), vol. 35, 1950

   1886
   no. 2, pp. 53-60, 2005, ISSN 0146-4833. [Online]. Avail- 1951

   1887
   able: https://citeseerx.ist.psu.edu/viewdoc/download? 1952

   1888
   doi=10.1.1.60.9560&rep=rep1&type=pdf
- 1889
   [53]
   Funcom, "Age of Conan," Jul. 2015. [Online]. Avail 1954

   1890
   able: https://web.archive.org/web/20150728133142/
   1955

   1891
   http://www.ageofconan.com/
   1956
- [54] A. Botta, A. Dainotti, and A. Pescapé, "A Tool 1957 for the Generation of Realistic Network Work- 1958 load for Emerging Networking Scenarios," *Computer* 1959 *Networks*, vol. 56, no. 15, pp. 3531–3547, 2012. [On- 1960 line]. Available: http://wpage.unina.it/a.botta/pub/ 1961
  COMNET\_WORKLOAD.pdf 1962
- [55] S. Ha, L. Le, I. Rhee, and L. Xu, "Impact of 1963
   Background Traffic on Performance of High-speed 1964

TCP Variant Protocols," *Comput. Netw.*, vol. 51, no. 7, pp. 1748–1762, May 2007. [Online]. Available: http://dx.doi.org/10.1016/j.comnet.2006.11.005

- [56] T. Dreibholz, NetPerfMeter Homepage, Jul. 2015. [Online]. Available: http://web.archive.org/web/20150728133903/ https://www.uni-due.de/~be0001/netperfmeter/
- [57] S. Hassayoun, J. R. Iyengar, and D. Ros, "Dynamic Window Coupling for Multipath Congestion Control," in Proceedings of the 19th IEEE International Conference on Network Protocols (ICNP), Oct. 2011, pp. 341– 352.
- [58] V. Cerf, V. Jacobson, N. Weaver, and J. Gettys, "BufferBloat: What's Wrong with the Internet?" ACM Queue, vol. 9, no. 12, pp. 10–20, Dec. 2011, ISSN 1542-7730. [Online]. Available: http://portal.acm.org/ ft\_gateway.cfm?id=2076798&type=pdf
- [59] J. Gettys, "Bufferbloat Dark Buffers in the Internet," Jan. 2011. [Online]. Available: https://www. bufferbloat.net/attachments/9/BufferBloat11.pdf
- [60] C. Raiciu, D. Niculescu, M. B. Braun, and M. J. Handley, "Opportunistic Mobility with Multipath TCP," in *Proceedings of the Sixth International* Workshop on MobiArch, ser. MobiArch '11. Bethesda, Maryland/U.S.A.: ACM, 2011, pp. 7–12, ISBN 978-1-4503-0740-6. [Online]. Available: http://nrg.cs.ucl.ac. uk/mptcp/mobility.pdf
- [61] C. Paasch, G. Detal, F. Duchêne, C. Raiciu, "Exploring Mobile/WiFi and O. Bonaventure, Handover with Multipath TCP. in Proceedthe ACM SIGCOMM Workshop on inas of CellularNetworks(CellNet),2012,31 pp. 36, ISBN 978-1-4503-1475-6. [Online]. Available: http://inl.info.ucl.ac.be/system/files/cell06-paasch.pdf
- [62] Y.-C. Chen, Y. Lim, R. J. Gibbens, E. M. Nahum, R. Khalili, and D. Towsley, "A Measurement-Based Study of MultiPath TCP Performance over Wireless Networks," in *Proceedings of the 13th ACM Internet Measurement Conference (IMC)*, Barcelona/Spain, Oct. 2013, pp. 455–468, ISBN 978-1-4503-1953-9. [Online]. Available: http://conferences.sigcomm.org/ imc/2013/papers/imc231-chenA.pdf
- [63] S. Deng, R. Netravali, A. Sivaraman, and H. Balakrishnan, "WiFi, LTE, or Both?: Measuring Multi-Homed Wireless Internet Performance," in *Proceedings of the* 14th ACM Internet Measurement Conference (IMC). Vancouver, British Columbia/Canada: ACM, 2014, pp. 181–194, ISBN 978-1-4503-3213-2. [Online]. Available: http://web.mit.edu/anirudh/www/imc086-dengA.pdf
- [64] A. Croitoru, D. Niculescu, and C. Raiciu, "Towards Wifi Mobility without Fast Handover," in *Proceedings* of the 12th USENIX Symposium on Networked Systems Design and Implementation (NSDI). Oakland, CA: USENIX Association, May 2015, pp. 219–234. [Online]. Available: https://www.usenix.org/conference/nsdi15/ technical-sessions/presentation/croitoru
- [65] K.-J. Grinnemo and A. Brunström, "A First Study on Using MPTCP to Reduce Latency for Cloud Based Mobile Applications," in Proceedings of the 6th IEEE International Workshop on Performance Evaluation of Communications in Distributed Systems and Web based Service Architectures (PEDISWESA), Jul. 2015.
- [66] L. Budzisz, J. Garcia, A. Brunstrom, and R. Ferrus, "A Taxonomy and Survey of SCTP Research," ACM Computing Surveys, vol. 44, no. 4, August 2012.

- [67] C. Casetti, G. Galante, and R. Greco, "Load balancing 2030 over multipaths using bandwidth-aware source schedul- 2031 ing," in *The 7th International Symposium on Wireless* 2032 *Personal Multimedia Communications (WPMC 2004)*, 2033
   Sep. 2004. 2034
- 1970 [68] M. Fiore and C. Casetti, "An adaptive transport 2035 protocol for balanced multihoming of real-time traf- 2036 fic," in *IEEE Global Telecommunications Conference* 2037 (*GLOBECOM 2005*), vol. 2, Nov. 2005, pp. 1091–1096. 2038
- 1974 [69] C. Xu, T. Liu, J. Guan, H. Zhang, and G. Muntean, 2039
  1975 "CMT-QA: Quality-aware Adaptive Concurrent Mul- 2040
  1976 tipath Data Transfer in Heterogeneous Wireless Net- 2041
  1977 works," *IEEE Transactions on Mobile Computing*, 2042
  1978 2012. 2043
- [70] S. Shailendra, R. Bhattacharjee, and S. Bose, "An im- 2044
  plementation of Min-Max optimization for multipath 2045
  SCTP through bandwidth estimation based resource 2046
  pooling technique," AEU International Journal of 2047
  Electronics and Communications, vol. 67, no. 3, pp. 2048
  246-249, 2013. 2049
- T. Dreibholz, [71]R. Seggelmann, M. Tüxen, and 2050 1985 E. P. Rathgeb, "Transmission Scheduling Opti- 2051 1986 mizations for Concurrent Multipath Transfer," 1987 in 2052 1988 Proceedings of the 8th International Workshop on 2053 Protocols for Future, Large-Scale and Diverse Network 2054 1989 Transports (PFLDNeT), vol. 8, Lancaster, Pennsyl- 2055 1990 vania/U.S.A., Nov. 2010, ISSN 2074-5168. [Online]. 2056 1991 https://www.wiwi.uni-due.de/fileadmin/ 2057 1992 Available: fileupload/I-TDR/SCTP/Paper/PFLDNeT2010.pdf 1993
- [72] G. Sarwar, P.-U. Tournoux, R. Boreli, and E. Lochin,
  "ecmt-sctp: Improving performance of multipath sctp
  with erasure coding over lossy links," in *Local Computer Networks (LCN), 2013 IEEE 38th Conference on*, Oct
  2013, pp. 476–483.
- [73] K.-J. Grinnemo, A. Brunstrom, and J. Cheng, "Using concurrent multipath transfer to improve the sctp startup behavior for pstn signaling traffic," in Advanced Information Networking and Applications Workshops (WAINA), 2014 28th International Conference on, May 2014, pp. 772–778.
- [74] D. Wischik, C. Raiciu, A. Greenhalgh, and M. Han-2005 dley, "Design, Implementation and Evaluation of 2006 Congestion Control for Multipath TCP," in Pro-2007 ceedings of the 8th USENIX Conference on Net-2008 worked Systems Design and Implementation (NSDI), 2009 Boston, Massachusetts/U.S.A., Mar. 2011, pp. 99–112. 2010 [Online]. Available: https://www.eecs.berkeley.edu/ 2011  $\sim$ sylvia/cs268-2014/papers/mptcp.pdf 2012
- 2013 [75] Q. Peng, A. Walid, J. Hwang, and S. H. Low,
  2014 "Multipath TCP: Analysis, Design, and Implementation," *IEEE/ACM Transactions on Networking*,
  2016 vol. PP, no. 99, p. 1, 2015. [Online]. Available:
  2017 http://arxiv.org/pdf/1308.3119
- [76] Y. Cao, M. Xu, and X. Fu, "Delay-based Congestion Control for Multipath TCP," in *Proceedings of the 20th IEEE International Conference on Network Protocols (ICNP)*, Austin, Texas/U.S.A., Nov. 2012, pp. 1–10, ISBN 978-1-4673-2445-8. [Online].
  Available: http://www.net.informatik.uni-goettingen.
  de/publications/1832/ICNP12-MPTCP.pdf
- 2025 [77] S. Ferlin, T. Dreibholz, and Özgü Alay, "Multi2026 Path Transport over Heterogeneous Wireless Net2027 works: Does it really pay off?" in Proceed2028 ings of the IEEE Global Communications Con2029 ference (GLOBECOM), Austin, Texas/U.S.A., Dec.

2014, pp. 4807–4813, ISBN 978-1-4799-3512-3. [Online]. Available: https://www.simula.no/sites/www.simula. no/files/publications/Simula.simula.2884.pdf

- [78] D. Wischik, M. Handley, and M. B. Braun, "The Resource Pooling Principle," ACM SIGCOMM Computer Communication Review (CCR), vol. 38, no. 5, pp. 47–52, Oct. 2008, ISSN 0146-4833. [Online]. Available: https://citeseerx.ist.psu.edu/viewdoc/download? doi=10.1.1.161.8858&rep=rep1&type=pdf
- [79] M. Kheirkhah, I. Wakeman, and G. Parisis, "Short vs. long flows: A battle that both can win," in Proceedings of the 2015 ACM Conference on Special Interest Group on Data Communication, ser. SIGCOMM '15. New York, NY, USA: ACM, 2015, pp. 349–350. [Online]. Available: http://doi.acm.org/ 10.1145/2785956.2790018
- [80] V. Sharma, S. Kalyanaraman, K. Kar, K. K. Ramakrishnan, and V. Subramanian, "MPLOT: A Transport Protocol Exploiting Multipath Diversity using Erasure Codes," in *Proceedings of the 27th IEEE INFOCOM*, Phoenix, Arizona/U.S.A., Apr. 2008. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/download? doi=10.1.1.115.280&rep=rep1&type=pdf
- [81] M. Kurant, "Exploiting the Path Propagation Time Differences in Multipath Transmission with FEC," *Journal on Selected Areas in Communications (JSAC)*, vol. 29, no. 5, pp. 1021–1031, May 2011. [Online]. Available: http://arxiv.org/pdf/0901.1479.pdf